Unmanned Aerial Vehicles Applications in Geoinformatics

Lead Guest Editor: Kamal Jain Guest Editors: Ram Avtar, Mozhdeh Shahbazi, and Xuan Zhu



Unmanned Aerial Vehicles Applications in Geoinformatics

Advances in Civil Engineering

Unmanned Aerial Vehicles Applications in Geoinformatics

Lead Guest Editor: Kamal Jain Guest Editors: Ram Avtar, Mozhdeh Shahbazi, and Xuan Zhu

Copyright © 2023 Hindawi Limited. All rights reserved.

This is a special issue published in "Advances in Civil Engineering." All articles are open access articles distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Chief Editor

Cumaraswamy Vipulanandan, USA

Associate Editors

Chiara Bedon (D, Italy Constantin Chalioris (D), Greece Ghassan Chehab (D, Lebanon Ottavia Corbi, Italy Mohamed ElGawady (D, USA) Husnain Haider 🕞, Saudi Arabia Jian Ji 🕞, China Jiang Jin 🕞, China Shazim A. Memon (D, Kazakhstan Hossein Moayedi D, Vietnam Sanjay Nimbalkar, Australia Giuseppe Oliveto (D), Italy Alessandro Palmeri (D, United Kingdom Arnaud Perrot (D), France Hugo Rodrigues (D, Portugal Victor Yepes (D), Spain Xianbo Zhao 🝺, Australia

Academic Editors

José A.F.O. Correia, Portugal Glenda Abate, Italy Khalid Abdel-Rahman (D, Germany Ali Mardani Aghabaglou, Turkey José Aguiar 🝺, Portugal Afaq Ahmad 🕞, Pakistan Muhammad Riaz Ahmad (D), Hong Kong Hashim M.N. Al-Madani (D, Bahrain Luigi Aldieri (D, Italy Angelo Aloisio 🕞, Italy Maria Cruz Alonso, Spain Filipe Amarante dos Santos (D, Portugal Serji N. Amirkhanian, USA Eleftherios K. Anastasiou (D), Greece Panagiotis Ch. Anastasopoulos (D, USA Mohamed Moafak Arbili (D, Iraq Farhad Aslani 🕞, Australia Siva Avudaiappan (D, Chile Ozgur BASKAN (D, Turkey Adewumi Babafemi, Nigeria Morteza Bagherpour, Turkey Qingsheng Bai (D, Germany Nicola Baldo (D, Italy Daniele Baraldi (D, Italy

Eva Barreira (D, Portugal Emilio Bastidas-Arteaga (D), France Rita Bento, Portugal Rafael Bergillos (D, Spain Han-bing Bian (D, China Xia Bian 🕞, China Husevin Bilgin (D, Albania Giovanni Biondi (D, Italy Hugo C. Biscaia (D, Portugal Rahul Biswas (D), India Edén Bojórquez (D, Mexico Giosuè Boscato (D, Italy Melina Bosco (D), Italy Jorge Branco (D, Portugal Bruno Briseghella (D, China Brian M. Broderick, Ireland Emanuele Brunesi (D), Italy Quoc-Bao Bui (D, Vietnam Tan-Trung Bui (D, France Nicola Buratti, Italy Gaochuang Cai, France Gladis Camarini (D), Brazil Alberto Campisano (D, Italy Qi Cao, China Qixin Cao, China Iacopo Carnacina (D, Italy Alessio Cascardi, Italy Paolo Castaldo D, Italy Nicola Cavalagli (D, Italy Liborio Cavaleri (D, Italy Anush Chandrappa (D), United Kingdom Wen-Shao Chang D, United Kingdom Muhammad Tariq Amin Chaudhary, Kuwait Po-Han Chen (D, Taiwan Qian Chen (D, China Wei Tong Chen (D, Taiwan Qixiu Cheng, Hong Kong Zhanbo Cheng, United Kingdom Nicholas Chileshe, Australia Prinya Chindaprasirt D, Thailand Corrado Chisari (D, United Kingdom Se Jin Choi (D), Republic of Korea Heap-Yih Chong D, Australia S.H. Chu D, USA Ting-Xiang Chu 🕞, China

Zhaofei Chu (D, China Wonseok Chung D, Republic of Korea Donato Ciampa (D, Italy Gian Paolo Cimellaro, Italy Francesco Colangelo, Italy Romulus Costache (D, Romania Liviu-Adrian Cotfas (D, Romania Antonio Maria D'Altri, Italy Bruno Dal Lago (D, Italy Amos Darko 🝺, Hong Kong Arka Jyoti Das 🕞, India Dario De Domenico (D, Italy Gianmarco De Felice D, Italy Stefano De Miranda (D, Italy Maria T. De Risi (D, Italy Tayfun Dede, Turkey Sadik O. Degertekin (D, Turkey) Camelia Delcea (D, Romania Cristoforo Demartino, China Giuseppe Di Filippo 🕞, Italy Luigi Di Sarno, Italy Fabio Di Trapani (D, Italy Aboelkasim Diab (D, Egypt Thi My Dung Do, Vietnam Giulio Dondi (D, Italy Jiangfeng Dong (D, China Chao Dou D, China Mario D'Aniello (D, Italy Jingtao Du D, China Ahmed Elghazouli, United Kingdom Francesco Fabbrocino (D, Italy Flora Faleschini (D, Italy Dingqiang Fan, Hong Kong Xueping Fan, China Qian Fang D, China Salar Farahmand-Tabar (D, Iran Ilenia Farina, Italy Roberto Fedele, Italy Guang-Liang Feng (D), China Luigi Fenu (D, Italy Tiago Ferreira (D, Portugal Marco Filippo Ferrotto, Italy Antonio Formisano (D, Italy Guoyang Fu, Australia Stefano Galassi (D, Italy

Junfeng Gao (D), China Meng Gao (D, China Giovanni Garcea (D, Italy Enrique García-Macías, Spain Emilio García-Taengua D, United Kingdom DongDong Ge (D, USA Khaled Ghaedi, Malaysia Khaled Ghaedi (D, Malaysia Gian Felice Giaccu, Italy Agathoklis Giaralis (D, United Kingdom Ravindran Gobinath, India Rodrigo Gonçalves, Portugal Peilin Gong (D), China Belén González-Fonteboa (D, Spain Salvatore Grasso (D, Italy Fan Gu, USA Erhan Güneyisi (D, Turkey Esra Mete Güneyisi, Turkey Pingye Guo (D, China Ankit Gupta (D, India Federico Gusella (D, Italy Kemal Hacıefendioğlu, Turkey Jianyong Han (D, China Song Han (D, China Asad Hanif (D, Macau Hadi Hasanzadehshooiili (D, Canada Mostafa Fahmi Hassanein, Egypt Amir Ahmad Hedayat (D, Iran Khandaker Hossain (D), Canada Zahid Hossain (D, USA Chao Hou, China Biao Hu, China Jiang Hu^(D), China Xiaodong Hu, China Lei Huang (D, China Cun Hui (D, China Bon-Gang Hwang, Singapore Jijo James (D), India Abbas Fadhil Jasim 🝺, Iraq Ahad Javanmardi (D, China Krishnan Prabhakan Jaya, India Dong-Sheng Jeng D, Australia Han-Yong Jeon, Republic of Korea Pengjiao Jia, China Shaohua Jiang (D), China

MOUSTAFA KASSEM (D), Malaysia Mosbeh Kaloop (D, Egypt Shankar Karuppannan 🕞, Ethiopia John Kechagias (D), Greece Mohammad Khajehzadeh (D, Iran Afzal Husain Khan D, Saudi Arabia Mehran Khan (D, Hong Kong Manoj Khandelwal, Australia Jin Kook Kim (D), Republic of Korea Woosuk Kim (D), Republic of Korea Vaclav Koci (D, Czech Republic Loke Kok Foong, Vietnam Hailing Kong D, China Leonidas Alexandros Kouris (b), Greece Kyriakos Kourousis (D, Ireland Moacir Kripka (D, Brazil Anupam Kumar, The Netherlands Emma La Malfa Ribolla, Czech Republic Ali Lakirouhani 厄, Iran Angus C. C. Lam, China Thanh Quang Khai Lam 🕞, Vietnam Luciano Lamberti, Italy Andreas Lampropoulos (b), United Kingdom Raffaele Landolfo, Italy Massimo Latour (D, Italy Bang Yeon Lee (D), Republic of Korea Eul-Bum Lee (D), Republic of Korea Zhen Lei 🕞, Canada Leonardo Leonetti D, Italy Chun-Qing Li 🝺, Australia Dongsheng Li (D, China Gen Li, China Jiale Li 🕞, China Minghui Li, China Qingchao Li 🝺, China Shuang Yang Li D, China Sunwei Li (D, Hong Kong Yajun Li 🕞, China Shun Liang (D, China Francesco Liguori (D, Italy Jae-Han Lim 🝺, Republic of Korea Jia-Rui Lin 🕞, China Kun Lin 🕞, China Shibin Lin, China

Tzu-Kang Lin (D, Taiwan Yu-Cheng Lin D, Taiwan Hexu Liu, USA Jian Lin Liu 🝺, China Xiaoli Liu D, China Xuemei Liu (D, Australia Zaobao Liu 🝺, China Zhuang-Zhuang Liu, China Diego Lopez-Garcia (D, Chile Cristiano Loss (D), Canada Lyan-Ywan Lu (D, Taiwan Jin Luo D, USA Yanbin Luo (D, China Jianjun Ma D, China Junwei Ma D, China Tian-Shou Ma, China Zhongguo John Ma (D, USA) Maria Macchiaroli, Italy Domenico Magisano, Italy Reza Mahinroosta, Australia Yann Malecot (D, France Prabhat Kumar Mandal (D, India John Mander, USA Iman Mansouri, Iran André Dias Martins, Portugal Domagoj Matesan (D, Croatia Jose Matos, Portugal Vasant Matsagar (D), India Claudio Mazzotti D, Italy Ahmed Mebarki (D, France Gang Mei (b), China Kasim Mermerdas, Turkey Giovanni Minafò (D, Italy Masoomeh Mirrashid (D, Iran Abbas Mohajerani (D, Australia Fadzli Mohamed Nazri (D, Malaysia Fabrizio Mollaioli (D, Italy Rosario Montuori (D, Italy H. Naderpour (D, Iran Hassan Nasir (D, Pakistan Hossein Nassiraei (D, Iran Satheeskumar Navaratnam (D, Australia Ignacio J. Navarro (D, Spain Ashish Kumar Nayak (D, India Behzad Nematollahi D, Australia

Chayut Ngamkhanong (D, Thailand Trung Ngo, Australia Tengfei Nian, China Mehdi Nikoo 🕞, Canada Youjun Ning D, China Olugbenga Timo Oladinrin (D, United Kingdom Oladimeji Benedict Olalusi, South Africa Timothy O. Olawumi 🕞, Hong Kong Alejandro Orfila (D, Spain Maurizio Orlando (D, Italy Siti Aminah Osman, Malaysia Walid Oueslati (D, Tunisia SUVASH PAUL (D, Bangladesh John-Paris Pantouvakis (D, Greece Fabrizio Paolacci (D, Italy Giuseppina Pappalardo (D), Italy Fulvio Parisi , Italy Dimitrios G. Pavlou (D, Norway) Daniele Pellegrini (D, Italy Gatheeshgar Perampalam (D), United Kingdom Daniele Perrone (D, Italy Giuseppe Piccardo (D, Italy Vagelis Plevris (D), Qatar Andrea Pranno (D), Italy Adolfo Preciado (D, Mexico Chongchong Qi (D, China) Yu Qian, USA Ying Qin 🝺, China Giuseppe Quaranta (D), Italy Krishanu ROY (D, New Zealand Vlastimir Radonjanin, Serbia Carlo Rainieri (D, Italy Rahul V. Ralegaonkar, India Raizal Saifulnaz Muhammad Rashid, Malaysia Alessandro Rasulo (D), Italy Chonghong Ren (D, China Qing-Xin Ren, China Dimitris Rizos (D, USA) Geoffrey W. Rodgers (D, New Zealand Pier Paolo Rossi, Italy Nicola Ruggieri (D, Italy JUNLONG SHANG, Singapore

Nikhil Saboo, India Anna Saetta, Italy Juan Sagaseta (D, United Kingdom Timo Saksala, Finland Mostafa Salari, Canada Ginevra Salerno (D), Italy Evangelos J. Sapountzakis (D), Greece Vassilis Sarhosis (D), United Kingdom Navaratnarajah Sathiparan (D, Sri Lanka Fabrizio Scozzese (D), Italy Halil Sezen D, USA Payam Shafigh (D, Malaysia M. Shahria Alam, Canada Yi Shan, China Hussein Sharaf, Iraq Mostafa Sharifzadeh, Australia Sanjay Kumar Shukla, Australia Amir Si Larbi (D), France Okan Sirin (D, Qatar Piotr Smarzewski (D, Poland Francesca Sollecito (D, Italy Rui Song D, China Tian-Yi Song, Australia Flavio Stochino (D), Italy Mayank Sukhija (D, USA Piti Sukontasukkul (D, Thailand Jianping Sun, Singapore Xiao Sun D, China T. Tafsirojjaman (D, Australia Fujiao Tang (D, China Patrick W.C. Tang D, Australia Zhi Cheng Tang (D, China Weerachart Tangchirapat (D, Thailand Xiaxin Tao, China Piergiorgio Tataranni D, Italy Elisabete Teixeira (D, Portugal Jorge Iván Tobón D, Colombia Jing-Zhong Tong, China Francesco Trentadue (D, Italy Antonello Troncone, Italy Majbah Uddin (D, USA Tarig Umar (D, United Kingdom) Muahmmad Usman, United Kingdom Muhammad Usman (D, Pakistan Mucteba Uysal (D, Turkey

Ilaria Venanzi (D, Italy Castorina S. Vieira (D, Portugal Valeria Vignali (D, Italy Claudia Vitone D, Italy Liwei WEN 🝺, China Chunfeng Wan (D, China Hua-Ping Wan, China Roman Wan-Wendner 🝺, Austria Chaohui Wang (D, China Hao Wang D, USA Shiming Wang (D, China Wayne Yu Wang 🕞, United Kingdom Wen-Da Wang, China Xing Wang D, China Xiuling Wang (D, China Zhenjun Wang 🝺, China Xin-Jiang Wei 🕞, China Tao Wen 厄, China Weiping Wen (D, China) Lei Weng D, China Chao Wu (D, United Kingdom Jiangyu Wu, China Wangjie Wu D, China Wenbing Wu (D, China) Zhixing Xiao, China Gang Xu, China Jian Xu 🕞, China Panpan (D), China Rongchao Xu (D, China HE YONGLIANG, China Michael Yam, Hong Kong Hailu Yang 🝺, China Xu-Xu Yang D, China Hui Yao (D), China Xinyu Ye_D, China Zhoujing Ye, China Gürol Yildirim (D, Turkey Dawei Yin D, China Doo-Yeol Yoo D, Republic of Korea Zhanping You 🝺, USA Afshar A. Yousefi 🕞, Iran Xinbao Yu D, USA Dongdong Yuan (D, China) Geun Y. Yun (D, Republic of Korea

Hyun-Do Yun D, Republic of Korea Cemal YİĞİT D, Turkey Paolo Zampieri, Italy Giulio Zani 🕞, Italy Mariano Angelo Zanini (D, Italy Zhixiong Zeng (D), Hong Kong Mustafa Zeybek, Turkey Henglong Zhang (D), China Jiupeng Zhang, China Tingting Zhang (D, China Zengping Zhang, China Zetian Zhang (D, China Zhigang Zhang (D, China Zhipeng Zhao (D, Japan Jun Zhao 🕞, China Annan Zhou (D, Australia Jia-wen Zhou (D, China Hai-Tao Zhu (D, China Peng Zhu (D, China QuanJie Zhu 🕞, China Wenjun Zhu^(D), China Marco Zucca, Italy Haoran Zuo, Australia Junging Zuo (D, China Robert Černý D, Czech Republic Süleyman İpek (D, Turkey

Contents

Uncrewed Aerial Systems in Water Resource Management and Monitoring: A Review of Sensors, Applications, Software, and Issues

Vishal Mishra (D), Ram Avtar (D), A. P. Prathiba (D), Prabuddh Kumar Mishra (D), Anuj Tiwari (D), Surendra Kumar Sharma (D), Chandra Has Singh (D), Bankim Chandra Yadav (D), and Kamal Jain (D) Review Article (28 pages), Article ID 3544724, Volume 2023 (2023)

Rainfall & Seismological Dump Slope Stability Analysis on Active Mine Waste Dump Slope with UAV Shirshendu Layek (), Vasanta Govind Kumar Villuri (), Radhakanta Koner (), and Kapoor Chand () Research Article (15 pages), Article ID 5858400, Volume 2022 (2022)



Review Article

Uncrewed Aerial Systems in Water Resource Management and Monitoring: A Review of Sensors, Applications, Software, and Issues

Vishal Mishra^(D),¹ Ram Avtar^(D),² A. P. Prathiba^(D),¹ Prabuddh Kumar Mishra^(D),³ Anuj Tiwari^(D),⁴ Surendra Kumar Sharma^(D),⁵ Chandra Has Singh^(D),¹ Bankim Chandra Yadav^(D),¹ and Kamal Jain^(D)

¹Department of Civil Engineering, Indian Institute of Technology Roorkee, Roorkee-Haridwar Highway, Roorkee, Uttarakhand 247667, India

²Faculty of Environmental Earth Science, Hokkaido University, Sapporo, Japan

³Department of Geography, Shivaji College, University of Delhi, New Delhi 110027, Delhi, India

⁴Discovery Partners Institute, University of Illinois at Chicago, IL, Chicago 60607, USA

⁵Indian Institute of Remote Sensing, Kalidas Road, Dehradun 248001, India

Correspondence should be addressed to Vishal Mishra; vmishra1@ce.iitr.ac.in

Received 1 March 2022; Revised 27 August 2022; Accepted 27 September 2022; Published 6 February 2023

Academic Editor: Asad Hanif

Copyright © 2023 Vishal Mishra et al. This is an open access article distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Uncrewed aerial systems (UASs) are becoming very popular in the domain of water resource mapping and management (WRMM). Being a cheaper and quicker option capable of providing high temporal and spatial resolution data, UAS has become a much sought-after platform for remote sensing. Still, their application in the field is in its early stage. This paper encompasses basic concepts of UAS, different payloads and sensor technologies available, various methodologies for its application in WRMM, different software available, and challenges associated with them, thus presenting a comprehensive review of multiple applications of UAS in different sub-domains of water resources. From cryosphere, rivers and lakes, and coastal areas to sub-surface water, as well as from water quality to wastewater management, the authors have discussed various applications of uAS in WRMM. Also, the future scope of the UAS in WRMM has been discussed.

1. Introduction

Water is the dispenser of life on this Earth [1], and almost three-quarters of the Earth's surface are covered with water. About 2.5% of the planet's available water is stored in the form of fresh water in the form of rivers, lakes, glaciers, polar ice caps, groundwater, soil moisture, water vapour, and many other forms. The distribution of water resources is heterogeneous both spatially and temporally. Therefore, continuous monitoring of water resources becomes necessary for water resource management as it is vulnerable to adverse consequences of anthropogenic activities. The UAS technology is particularly suitable for qualitative and quantitative analysis such as mapping and monitoring of dynamic components of the water cycle [2] such as soil moisture, runoff, evapotranspiration, snow cover, and so on, along with water structures and water-related disasters.

UASs have emerged as a viable option to fill many gaps between spaceborne and terrestrial in situ observation technologies [3], offering (1) availability of high spatial resolution at lower cost with high temporal resolution; (2) subjective time of sampling by the operator; and (3) payload flexibility [4]. UAS are, therefore, inexpensive [5, 6], versatile, and safer, which can be very helpful in water resource management and monitoring. Many applications in the domain of water resources demand ultra-high-resolution data that satellites cannot provide, and that can be met by UAS [7]. The majority of water problems lie in developing countries with poor access and a lack of data. In such cases, UAS can be of maximum use in monitoring and managing water resources. The UAS data can also help us to broaden our understanding of how global challenges such as climate change and population growth affect our water resources at the local level. Water resources are running out, and real-time water resource management requires increasingly accurate data on water, soil, and vegetation conditions compared to other monitoring applications. Many applications require ultra-high-resolution data that satellites cannot provide and that UAS can fulfil [3, 4]. The WRMM sector is evolving day by day, adding new applications of UAS. From detecting sealevel changes over time [8], lake level changes [9], estimating water storage [10], monitoring riparian systems [11, 12], flood assessment [13-17] to mapping glaciers [18], drones have been used. Recent developments have made it possible to identify the location and spatiotemporal changes in groundwater storage [19]. Thus, there are a plethora of applications and methods based on UAS data that are being studied. New drone-based solutions (e.g., DOWSE [19]) are being developed in sub-domains of WRMM to increase the data collection efficiency and render services much faster. There have been many reviews on the application of UAVs in water resources [2, 18, 20-25]. Some of these reviews are old or mainly focused on only a component of water resource monitoring. These were very focused and lacked an overall synopsis of the WRMM domain.

1.1. Objective. The main focus of this study is to assess the present state of research, opportunities, and challenges to achieve greater application of UASs in water resource monitoring and their use in their management. This is addressed with the following questions:

- (i) What are the different UAS sensors that currently exist?
- (ii) How can various UAS observations be used in WRMM?
- (iii) Which new WRMM applications have been innovated that can be put into practice?
- (iv) What are the challenges of UAS technology and the prospects for their implementation?

Section 1 introduces the objective of this paper. Section 2 highlights the historical progression of UAS and a brief account of sensors that are used in WRMM. Section 3 discusses the different applications of UAS in WRMM. We have divided the applications of UAS in WRMM into two broad categories: surface water resources and sub-surface water resources. Section 4 deals with the issues in the application of UAS in different sectors of water resources. In Section 5, we have tried to give a synopsis of software available for structure from motion (SfM)-based processing of UAS-based images. Section 6 discusses the challenges and Section 7 delves into the prospects of this technology. Section 8 summarizes the paper and is titled as conclusion.

Given the rapid proliferation of UAS applications in WRMM, a full investigation of the current state of the art is required to provide a clearer view and encourage further advances. A detailed analysis of existing work is essential for the continuous improvement of UAS application in WRMM, especially for researchers who want to enter the field. For this reason, this article provides a detailed overview of recent breakthroughs in UAS technologies and applications, focusing on UAS photogrammetry and UAS remote sensing. The area of water resources is itself very vast. Nevertheless, the authors have tried to provide an overview of sensors, applications, software, problems, and the future scope of UAS from the water resource point of view. Future research opportunities are also discussed.

2. UAS: Origin, Types, and Sensors

2.1. UAS: Definition and Characteristics. Uncrewed aerial systems consist of crewless aircraft and the equipment to control them remotely. The aircraft or vehicle is also referred to as an uncrewed aerial vehicle or drone or remotely piloted aircraft or uncrewed autonomous vehicle.

A UAS is a system of systems, i.e., a collection of complementary technologies brought together to achieve a specific goal, and therefore there are many different types of UAS in the market today: it can be said that there is one for each technical combination [26]. At the highest level of UAS technology, three key UAS components are typically identified: the uncrewed aerial vehicle, the ground control station, and the communication data link. A communication data link is a connection between the UAV and the ground control station. The ground control station can be explained as the controlling element responsible for the safe and automated flight of the UAV. These are immobile or mobile hardware/software devices used to monitor and control the uncrewed aerial vehicle [26]. Table 1 shows the advantages and disadvantages of UAS.

The development of UAS was initially driven purely by military applications; as with many other areas of technology, civilian use tends to take control once it has proven itself in the military field. If we consider the basic characteristics of UAVs that they are vehicles that generate aerodynamic lift and/or have some degree of control, it can be said that the kite is the first UAV.

Different platforms for remote sensing or acquiring geodata have many advantages and disadvantages. UAV supersedes in terms of resolution, flexibility, and, to a certain extent, cost-effectiveness. These are listed in Table 1. Based on the review, simplified taxonomy (see Table 2) has been provided on the UAS-based studies applied to the WRMM.

2.2. Classification of UAVs. The classification of UAVs differs from place to place and depends on various parameters. UAVs are divided into three main classes by UVS International [28]. These are

- (i) Strategic UAVs.
- (ii) Tactical UAVs.
- (iii) Special task UAVs.

Tactical UAVs have low endurance and operate from a few meters to five kilometers from the Earth's surface. Strategic ones have a two to 1-day lifespan and operate above 20,000 m altitude. Special-duty UAVs include lethal systems and deception systems.

According to the DGCA (Government of India) [29], UAVs are classified into four classes according to their maximum take-off weight (MTOW), as indicated in Table 3. UAVs can also be classified based on wings and rotors, as shown in Figure 1. Table 4 provides the pros and cons of different types of UAVs.

2.3. Payloads and Sensors. Being a remote platform system, UAS can be loaded with various sensors and instruments based on application requirements. Based on the literature, these UAV payloads used for WRMM can be classified in three classes, i.e., active sensors, passive sensors, and samplers (Figure 2). These can also be classified into imaging sensors, non-imaging sensors, and samplers as shown in Table 5. Remote sensing data can be very instrumental in developing hydrological models. The remote sensors and their data provide the capability to estimate many governing variables of the hydrological cycle, compatible with many hydrological applications. There are also many limitations and issues in fitting these remote sensing payloads into a predefined aircraft system, but development continues to adapt in-house payload systems for different sensors. Different types of UAS-based sensors have been discussed in the subsequent sections.

2.3.1. RGB, NIR, and Multi-Spectral Cameras/Optical Sensors. UAS equipped with inexpensive and lightweight RGB cameras have become standard for remote sensing and photogrammetric research. The simplest and least expensive sensors to deploy are optical ones, and when used appropriately, they can produce high-quality data for WRM [31]. Commercially available lightweight multi-spectral cameras (sensors) for tiny UAVs are now available. High-end multispectral sensors are helpful for surveillance because they provide high-quality measurements regardless of lighting (Table 5). These multi-spectral sensors are ideal for a wide range of applications due to their high ground sensing distance down to the centimeter range and their affordable cost. The disadvantage of current sensors of this type is that they are not properly tuned to aquatic applications [32]. When categorizing UAV images, the presence of an additional NIR band with RGB data is beneficial [33]. These sensors have been used in the majority of the UAV-based WRMM studies. This includes water body mapping, river bathymetry determination [34], flood mapping, riparian vegetation mapping [12], water quality mapping, and estimating soil water content [35]. Computer vision techniques find much utilization in deriving information from UAVacquired RGB data. Even video captured from cameras has been utilized in WRM applications [34].

2.3.2. Hyperspectral Cameras. Many aspects of water resource management have benefited from hyperspectral imagery. Many such hyperspectral sensors have been developed that are compatible with UAVs [27, 36-38]. The application of UAS-based hyperspectral sensors has increased due to their high degree of automation and rapid manoeuvrability. These UAV-compatible spectral sensors are classified as point, push broom, multi-camera, sequential snapshot (multi-point, filter-on-chip, and labelled RGB), and spatial-spectral sensors [39]. The common disadvantages of these sensors are (1) exorbitant prices, (2) sensor dimensions, (3) requirement for specialized software, and (4) low signal-to-noise ratio. Vibration and flight motion also affect the push broom sensors [40]. The UAV-based hyperspectral sensors found application in water quality [41, 42] and mapping of water infiltration rate [43].

2.3.3. Thermal Cameras. Thermal infrared (TIR) sensors can be used to assess soil surface moisture parameters, estimate spatial and temporal scale energy exchange and vegetation cover, and estimate an area's evapotranspiration. TIR data can be used to determine the water content of vegetation and is therefore very resourceful in adjusting local (precision farming) and global (sustainable management of water resources) irrigation water levels. These are very helpful in identifying flow features [52]. Thermal imagers equipped with microbolometer sensors mounted on UAS can provide thermal imaging. The advantages of these sensors are that no validation is required to measure relative temperatures and their lower cost. TIR cameras provide only a single band with very low image resolution [53]. The low sensor resolution, together with a narrower field of view (FOV), complicates the applicability of structure from motion algorithms with TIR images and potentially limits the scale at which UAS-based TIR imaging is suitable [52]. The disadvantage of these sensors is that they require radiometric corrections and are plagued by temperature drift problems. Another issue is the need for expertise in interpreting data products. Thermal radiation from the emitters at the waterland boundary can cause errors. UAS-based TIR remote sensing can be used to map and detect discrete cold or warm water inputs into river channels. These sensors are useful for irrigation monitoring when (i) pairing them with simultaneous capture of visible and near-infrared imagery; (ii) a geometric correction is made to overlay with other images; and (iii) a radiometric correction is applied to account for the drift of the thermal sensors as well as the influence of the atmosphere on the observed temperature [54]. UAV-based thermal images have been used for estimating the soil water content [35, 44]as well as for monitoring water pollution [49-51]. Another application is estimating land surface temperature (LST), which can then be used to estimate evaporation, which is important for water resource management [45, 46]. Table 6 presents the different thermal sensors used in literature and their applications in WRMM.

Platforms	Advantage	Disadvantage
Satellite	(i) Wider coverage	(i) Low resolution (ii) Image acquisition timing
	(ii) Broad spectral capability	(iii) Weak coverage in some regions(iv) Sensitive to clouds
	(i) Large coverage with a single flight	(i) Expensive (for small projects)(ii) Image acquisition timing
Manned aircraft	(ii) High resolution	(i) Weather-dependent(ii) Sensitive to clouds
	(iii) Wide spectral capability	(iii) Not available in remote regions
UAV	 (i) Cost-effective for small projects (ii) Very high resolution (fixed-wing up to 2 cm/pixel; rotary: sub-millimeter) (iii) Because of the lower flight height, clouds do not affect the flight (iv) Positional accuracy 	 (i) Small coverage (ii) Regulations may restrict operations (iii) Sensitive to bad weather (iv) Difficult to reconstruct homogeneous areas (few tie points)
Terrestrial	 (i) Excellent positional accuracy (ii) Few data (only required) (iii) Very high resolution (iv) In situ data classification 	(i) Labour intensive(ii) Only line-of-sight(iii) Accessibility (some sites)

TABLE 1: Pros and cons of the existing remote sensing technologies [27].

TABLE 2: Taxonomy for operating UAV-based remote sensing systems in water resource applications.

Category	Sub-category
(A) Water resource	Surface sub-surface water structures
(B) Application	Reactive proactive passive
(C) Processing	Preprocessing segmentation regression classification 3D reconstruction
(D) Payload	Active passive samplers
(E) Platform	Fixed-wing single rotors multi-rotors lighter than air UAS

TABLE 3: Classification of UAVs based on their maximum take-off weight.

Category	Specification
Nano	Weight ≤ 250 g
Micro	$250 \mathrm{g} < \mathrm{weight} \le 2 \mathrm{kg}$
Mini	$2 \text{ kg} < \text{weight} \le 25 \text{ kg}$
Small	$25 \text{ kg} < \text{weight} \le 150 \text{ kg}$
Large	Weight > 150 kg

2.3.4. Radar and Synthetic Aperture Radar. Many UASbased radars are being developed [55] but suffer from bottlenecks at both hardware and software levels [56]. Bandini et al. [57] found in their study that radar is the most reliable sensor for measuring water levels compared to sonar and camera laser sensors. Synthetic aperture radar (SAR) interferometry is a powerful tool for terrain mapping [58]. In the literature, most UAV-based SAR systems operate in the X or Ka-band. The NASA-JPL (National Aeronautics and Space Administration-Jet Propulsion Laboratory) has designed and developed a L-band UAV-based SAR (UAVSAR) and developed UAVs for topographic mapping over the decade [58]. The UAVSAR data have been used to assess changes in water levels in wetlands due to tides [42], monitor the land subsidence and aquifer depletion [59-61], etc. [62]. Table 7 summarizes the characteristics of SAR sensors in relation to platforms. Ouchi [63] presented a nice

overview of UAV SAR sensors up to 2013. Ludeno et al. [55] described an experiment with a mounted micro-UAV radar system. Wu [64] demonstrated the development of UAV-based ground-penetrating radar for soil moisture mapping. UAV-based radars have also been used to estimate snow depth and density [65]. These sensors are also needed to be explored for WRMM studies.

2.3.5. Radiometer. UAV-based radiometers are suitable for regional or local applications for remote measurement of geophysical parameters, such as soil moisture (SM) or sea surface salinity (SSS) [66], and can be used to detect saltwater infiltration. These are less sensitive to atmospheric influences than satellite-based radiometers.

2.3.6. Gravimeters. Gravimeters are useful in modelling and estimating changes in water storage. Conventional gravimeters are expensive and have a high mass, making them unsuitable for UAV assembly. But lately, some UAV gravimeters are being developed [67–69]. UAV-based gravimetry can complement satellite-based gravity observations and can be beneficial in remote and transitional regions (coastal waters) where terrestrial gravity measurements are difficult and impractical. One such gravimeter is the gravimeter of a microelectromechanical system (MEMS). MEMS gravimeters can be mounted on UAVs [68]. They



FIGURE 1: UAV types.

TABLE 4: Classification of UAVs.

	Pros	Cons	Uses	
Multi-rotor (quad and	(i) Accessibility(ii) Ease of application(iii) VTOL and hoverflight	(i) Short flight times	Photography, simple photogrammetric	
nexacopters)	(iv) Good camera(v) Can operate in a confined area	(ii) Low payload capacity	applications, and video inspection	
	(i) High endurance	(i) No VTOL/hover		
Piece I and a	(ii) Coverage of large area	(ii) More challenging to fly, skilled training required	Aerial mapping, road, pipeline, and power line	
Fixed-wing	(iii) Fast flight speed	(iii) Costly(iv) Launch and recovery need a lot of space	inspection	
	(i) VTOL and hover flight	(i) More dangerous		
Single-rotor	(ii) Long endurance	(ii) Harder to fly, more training needed	Aerial laser scanning (ALS)	
2	(iii) Higher payload- carrying capacity	(iii) Expensive	- · ·	



FIGURE 2: Classification of UAV payloads for water resource management and monitoring.

TABLE 5: Potential UAS payloads/sensors for water resource monitoring and management.

Imaging sensors	Non-imaging sensors	Samplers
Multi-spectral camera	Gravimetric sensors	Water samplers
Infrared sensors	Electromagnetic induction sensors	_
Thermal sensors	Thermal profiler	
Hyperspectral sensors	Radiometers	
Microwave sensors		
Light detection and ranging		
Laser fluorosensors		
Magnetometers [30]		

Reference/ study	Sensor used	Range of sensor used	Application of thermal sensor/camera
[12]	ICI mirage 640	3.4 μm–5.1 μm	Computing river discharge
[35]	Zenmuse XT2-uncooled vox microbolometer	7.5 μm–13.5 μm	Predicting soil water content
[44]	ZENMUSE XT	7.5 μm–13.5 μm	Soil moisture retrieval
[45]	Optris PI 450 light weight infrared	$7.5 \mu m - 13 \mu m$	Estimating evaporation
[46]	Optris Pi 400	7.5 μm–13 μm	Estimating spatially distributed turbulent heat fluxes
[47]	FLIR A65	7.5 μm–13 μm	Measuring surface flow velocity
[48]	FLIR Tau2 324	$7.5 \mu m$ – $13.5 \mu m$	Monitoring water flux
[49]	Workswell Wiris 640 as	$7 \mu m - 14 \mu m$	Oil spill monitoring
[50]	FLIR thermal sensor	$8 \mu\text{m}$ – $14 \mu\text{m}$	E. coli pollution monitoring
[51]	FLIR Vue Pro R 640	7.5 μm–13.5 μm	Monitoring floating marine plastic litter

 TABLE 6: Different thermal sensors used for different WRMM applications.

TABLE 7: UAV SAR sensor characteristics as compared with airborne and satellite platforms (the sign "ü" means that the requirement is fully attended, "—" means partially attended, and "û" means that the requirement is not attended [62]).

Requirements	UAV	Satellite	Airborne
Resolution (high)	ü	—	ü
Precision (high)	ü	_	ü
Coverage	û	ü	_
Endurance	ü	û	—
Flexibility	ü	û	ü
Rapid deployment	ü	_	û
Low-complexity operation	ü	—	û
Low-complexity logistics	ü	_	û

have the advantage of being mass-producible, lightweight, and cheap. Robust field implementation of these UAV sensors is still pending.

2.3.7. LiDAR. UAS-based LiDAR (light detection and ranging) can be important in capturing high-resolution terrain information, which can help improve visualization and analysis. LiDAR has a distinctive advantage over traditional topographic survey techniques, namely, its ability to derive a more realistic, high-resolution digital elevation model (DEM). LiDAR point density (the number of points/ m^2) varies with flight speed [70]. Benefits of this payload include reduced susceptibility to environmental factors and direct geometry measurement. The disadvantages are that it is costly [71] and that the accuracy of the measurements can be affected if the vehicle is not properly stable. Another essential aspect that cannot be ignored is that water absorbs wavelengths commonly used for LiDAR [72]. The

phenomena of water volume scattering, water surface reflection and refraction, and turbidity also complicate data modelling. UAV-based LiDAR bathymetry can be useful for underwater object detection, 3D mapping of underwater topography, turbidity estimation, and river and coastal geomorphology and applications [73].

2.3.8. Laser Fluorosensors. Instruments that measure fluorescence with laser beams are called fluorescence LiDAR or laser fluorescence sensors or laser-induced fluorosensors. They are used to analyze physical (e.g., oil spill monitoring) and biological parameters of water bodies, such as turbidity and algae content. Laser fluorine sensors take advantage of the fact that certain substances, such as aromatic compounds in petroleum, absorb ultraviolet light and become electronically excited. This excitation initiates fluorescence emission. These have also been developed for UAVs [74, 75]. This fluorosensor application should not be confused with

the UAV application for measuring fluorescent tracers [76], which uses only RGB cameras. The application in WRMM is still in the development phase.

3. UAS Data Reduction Workflows for UAV Images/Sensors

Before discussing how UAS-based data are used for WRMM, it is essential to understand how data are derived from UAV imagery. Just like other remote sensing applications, UAVbased imagery can be used to derive mainly two categories of information: metric and thematic. Hence, the corresponding processes based on the output of images captured by UAV can be called UAV photogrammetry (a term coined by Eisenbeiss [77]) and UAV remote sensing, respectively.

The essential part of optical data acquisition is flight planning, regardless of the technology used. One major challenge is determining how to plan the flight path to ensure a complete and accurate survey of the study area with the least flight time. The camera is typically positioned in a fixed orientation, such as vertical or oblique, and the UAV-RS collects data either manually or using preprogrammed flight paths. Therefore, full and dense coverage is difficult to achieve, especially in urban areas. Using the initial scene reconstruction from nadir acquisition as a baseline to continually optimize the view and location is an interesting technique.

UAV photogrammetry is essentially a hybrid of aerial photogrammetry and close-range photogrammetry (CRP). Here the platform is in the air, but the data reduction follows the principles of CRP. UAV photogrammetry generally applies algorithms called structure from motion (SfM) data reduction algorithms. They involve the simultaneous determination of the (internal and external) geometry features as well as the 3D structure of the scene [79, 80]. This algorithm uses images captured by optical sensors and the positions of their exposure stations as their input, and their outputs are 3D point clouds and digital elevation models, and after ortho-rectification, they result in ortho-mosaics. Al-Kaff et al. [81] presented a comprehensive overview of structure from motion algorithms along with their applications and limitations. Conventional photogrammetric processing of UAS data is presented in Figure 3. The UAVbased 3D models are rich in detail and can therefore be used to obtain knowledge about the hydraulic parameters of waterways [82]. But UAV photogrammetry has its bottlenecks. Insufficient lighting and shadows are the sources of error with SfM products. In snowy areas, the SfM algorithm has difficulty processing due to a lack of contrast and very high reflectivity [18]. For bathymetric mapping applications, UAV-borne multi-media photogrammetry is applied as it involves light diffraction at the air-water interface [73].

UAV remote sensing is the branch of remote sensing that uses UAV as a platform to acquire various parameters about the objects or phenomena to be observed. UAV-based remote sensing uses the application of digital image processing, which exploits the physics associated with the interaction of radiation belonging to a specific range of the electromagnetic spectrum, including the optical range. Using the basic principles of remote sensing, the application of UAS-based data for mapping and characterizing water bodies can be made. Optical remote sensing has some limitations when it comes to mapping water. These include obstacles from other features, shadows on the water surface, changing water surface appearance due to sun-target-sensor geometry, and the dynamic morphology of water bodies [83, 84]. The near-infrared (NIR) and mid-infrared (MIR) regions of the electromagnetic spectrum, with wavelengths between 740 and 2500 nm, are best suited to distinguishing pure water from land [85].

4. UAS in Water Resource Monitoring and Management

There are many different methods for using UASs in WRMM. A standard methodology is to assess the feasibility of UAV surveillance after considering spatiotemporal coverage, acquisition parameters, data quality, legal issues, etc. A decision is taken on how UAS monitoring will be carried out after considering factors like cost and detection parameters, weather conditions, and accessibility of the study area. The firsthand acquisition is an image or a point cloud, or a sample. Then, this image, point cloud, or sample is further processed/ analyzed to obtain the primary data, which in turn undergoes secondary processing to provide results. The results are interpreted and displayed in the form of maps, charts, graphs, etc. online or offline. Figure 4 describes the general sequential process of decision making from problem definition to the operational aspect of the application.

Before starting the review, we will summarize the review papers (Table 8) directly or indirectly related to UAVs and their application for mapping and monitoring water resources in the table below. In this overview, we have tried to summarize all the developments that have been made in the mapping and monitoring of water resources. Some of these reviews are old or mainly focused on only a component of water resource monitoring. For our review purpose, we have divided the WRMM applications into three categories, namely, (I) surface water resources, (II) sub-surface water resources, and (III) irrigation and other water structures.

4.1. UAS for Surface Water Resources. There are different areas of WRMM where UASs are being used. Table 9 summarizes different applications as per different water bodies.

4.1.1. Mapping and Characterizing Water Bodies. The mapping of (inland) water bodies is essential for the mapping and monitoring of water resources. Sub-meter imagery captured by drones allows for more accurate delineation of water features, which is always desirable for scientists and policymakers. Another aspect is the improved detection of small bodies of water. These include flow tracing, channel bathymetry, and thermal characterization of aquatic ecosystems [155]. Using a thermal imaging camera can help to monitor the seasonal geothermal influence on the rivers [156]. UAV datasets, in synergy with other datasets, can be used to map bodies of water.



FIGURE 3: Conventional photogrammetric processing pipeline for UAS data [78].

D'Andrimont et al. [84] used hyperspatial and multi-source data for mapping bodies of water over a large area. The framework proposed in the study successfully handled the heterogeneity of different remote sensing platforms and detected 83% of the water bodies in the area. Fu et al. [104] used UAVs to map land use/land cover, particularly ponds, to assess ecosystem services provided by ponds in hilly areas. Harvey et al. [157] used calibrated thermal images obtained from UAS to study thermal lakes and streams. The study used these images and the Monte Carlo analysis to estimate a mean total heat loss at the surface. Another study [45] showed that UAVs could be used to estimate evaporation. In this study, the surface energy balance components were calculated using land surface temperature from UAS-based thermal imagery and used as input to the physically based two-source energy balance models. Kuhn et al. [158] attempted again to assess thermal heterogeneity via UAVbased imaging. UAS-based thermography has been used to monitor surface water contamination [159, 160]. UAVs have been used to study the shoreline and shoreline erosion of inland lakes and rivers [161-163].

4.1.2. Watershed Mapping and Monitoring. UAV photogrammetry using overlapping stereo images provides very detailed information about terrain, catchment areas, and networks. By providing a thorough understanding of the watershed status and changes over time, UAVs can be used to validate products from various existing and upcoming satellite missions. UAVs can be used to address the need for cross-watershed monitoring and assessment of the large geographic diversity in underground hydrological connections [31].

Templeton et al. [164] characterized the terrain attributes (elevation, slope, orientation, and upstream area) and plant species distribution in a catchment using UAV products supported by an environmental sensor network. The study analyzed the dynamics of energy and water fluxes in the watershed on different timescales (i.e., seasonal, monthly, and storm events) and provided insights into their spatial variations and their interconnection. Spence and Mengistu [165] applied UAS imagery to identify narrow intermittent streams. Pineux et al. [166] presented a UAS-based technique for quantifying the spatiotemporal distribution of erosion/deposition due to precipitation events. This technique can be used to study erosion in the watershed where other methods are too expensive, destructive, or timeconsuming. Argüello et al. [167] described a methodology for automatic river basin monitoring using a UAV-mounted multi-spectral sensor.

(1) River Mapping and Velocimetry. UAS remote sensing and photogrammetry products have many applications for studying river systems. These include bathymetric survey, topographic survey, grain size mapping, sediment transport path length, geomorphological change detection, habitat classification, restoration monitoring, vegetation mapping, etc. [168]. Monitoring river discharge is an essential task for water resource management. The study of river morphology is a crucial task in river management. River management facilities such as dikes and river walls play a vital role in flood control. UAVs can help map and monitor all of this. Due to the higher spatial resolution, unmanned aerial vehicles are very well suited for exploring small rivers and streams [169].

Although the major rivers can be mapped from satellite data, smaller rivers flowing through dense vegetation are obscured, making UAVs suitable for mapping small rivers [170]. UAV data are useful in deriving channel parameters, identifying hydromorphological features [171], and studying river dynamics [124] such as geomorphological changes due



FIGURE 4: The general sequential process of decision making from problem definition to the operational aspect of the application.

to flooding [125, 171, 172]. Casado et al. [171] used ANN to automatically identify different flow characteristics. Kubota et al. [173] proposed a river asset maintenance management system using UAS-derived three-dimensional point cloud data to solve river management asset problems and improve operational and maintenance efficiency. Zhao et al. [174] used UAV-derived data to collect channel parameters for rapid environmental flow assessment. The DEM derived from the UAV can, in turn, be used to create a flood depth model and other parameters (roughness indices) [26, 117] and thus used in bank erosion studies [112] and flood modelling [70, 109]. The method of precise aerial

Year	Review topic	Reference
2011	UAV application for environmental remote sensing	[86]
2014	UAVs for 3D mapping application	[78]
2016, 2017	UAV for hazards and disasters	[87, 88]
2016, 2018, 2020, 2021	UAVs for glaciology	[18, 22, 88, 89]
2019	UAV applications in urban storm water management	[90]
2017	Deep learning application for UAV	[91]
2018, 2019	UAV application for monitoring algal bloom	[24, 92]
2017	UAV hyperspectral sensors	[27]
2018	UAV-based spectroscopy	[39]
2020	Structure from motion algorithms for UAV mapping	[93]
2018	UAV for fluvial remote sensing	[94]
2019	UAV for water sampling	[20]
2021	UAV and satellite data synergy	[3]
2021	Accuracy of UAV mapping	[80]
2021	The role of UAS technology in natural resource management	[95]

TABLE 8: Topics discussed in the previous articles.

TABLE 9: The application of UAVs in WRMM.

Water body type	References
Lakes and reservoirs	[96-103]
Ponds	[104]
Alpine glacial lakes	[105]
Rivers and river basins	[52, 94, 106–125],
Wetlands	[33, 47, 101, 126–138]
Glaciers	[139–141]
Delta, ocean, and coastal regions	[8, 9, 59, 142–153]
Ice caps	[154]

photogrammetry [82, 175] can be applied to UAS imagery to derive geomatic products that can be used as input for flood modelling. UAS-based thermal imaging cameras have been used to monitor seasonal geothermal influence on the rivers [52]. Calle et al. [172] applied UAV photogrammetry to monitor short-lived river changes due to flooding. Langhammer and Vacková [112] used UAS to map the geomorphological effects of flooding with the object-based image analysis. Gebrehiwot et al. [14] applied a deep convolutional neural network to UAS imagery for flood extent mapping. This study achieved an accuracy of 95%. Hashemi-Beni and Gebrehiwot [17] integrated CNN and regiongrowing (RG) method for mapping existing floods using UAS imagery.

The relevant parameter of UAV surveys for rivers is the area coverage (longitude and latitude). From this source come the associated challenges. On narrow rivers, it is conceivable to fly just one line of flight over the middle of the river while the camera's field of view covers the entire width. However, when the river is too great for a single airline to span, round-trip back-and-forth flights are required, significantly reducing travel time. The line-of-sight restriction mentioned above limits the ability to fly along the river. Larger rivers need multiple flights [176].

Another challenging aspect of river mapping is the vegetation along the river. Mapping is complicated in fastflowing water, blocked waterways, or dense tree canopies. These trees can block GPS signals. Unfortunately, river conditions severely degrade any GPS signal, resulting in intermittent global position data. The UAS can help overcome these difficulties. Scherer et al. [177] developed a UAV for river exploration and mapping that can estimate position and recognize the river and obstacles. Some of these studies [170, 177] showed that GPS waypoints and previous maps were hardly or not at all required for the autonomous exploration of riverine environments.

Water level and water surface height can be derived from various UAV-based sensors. UAS-based photogrammetry represents a dynamic and non-intrusive approach to studying free-surface topography. Bandini et al. [57] determined levels with an accuracy of better than 57 cm using an integrated payload consisting of a camera-based laser distance sensor (CLDS), radar, and sonar. Water surface elevation data can significantly improve flood forecast models, advance our knowledge of how river geometry and hydraulic roughness affect WSE, and contribute in constructing assessment curves. The currently captured description of the high-resolution surface morphology can answer fundamental questions related to the nature of the free surface [178]. Ridolfi and Manciola [98] performed drone-based measurements of a lake and reported that the total mean error between estimated and actual water level measurements is about 0.05 m. Eltner et al. [179] used deep learning techniques (SegNet and FCNN) in combination with UAV photogrammetry for automatic water level measurement.

UAVs have been used to measure flow velocity. UASacquired images have been used in many studies to derive surface velocity [47, 47, 126, 127, 132, 133, 137, 180]. The studies mainly focused on evaluating the technique. Tauro et al. [127] compared results of UAS-based large-scale particle image velocimetry (LSPIV) with in situ measurements and attempted to estimate the impact of platform stability on the results. Koutalakis et al. [181] demonstrated flow velocity measurement by three image-based methods. Strelnikova et al. [133] analyzed photos of the area around the fishway entrance of a dam using UAS-based imagery under seed flow conditions. Pearce et al. [126] performed a sensitivity analysis for five different image velocimetry algorithms, namely, large-scale particle image velocimetry, Kanade–Lucas–Tomasi image velocimetry, large-scale particle tracking velocimetry, surface structure image velocimetry, and optical tracking velocimetry.

4.1.3. Riparian Vegetation Monitoring. The study and management of riparian vegetation is a part of water resource management as it influences various hydrological processes [182]. UAVs are generally used to survey riparian systems on a local scale [12]. A trend was noted in the studies that UAVs are used to study the features associated with the diverse species composition of the riparian ecosystem. Dunford et al. [120] performed one of the first studies on applying UAS to monitor riparian vegetation by implementing pixel-based and object-oriented classification. One of the reviews [21] summarized that the number of studies using UAS to investigate the bank system increased after 2010. Husson et al. [11] concluded in their research that the sub-decimeter resolution of UAV products can be beneficial in mapping river and lake vegetation at the species level. Michez et al. [183] used hyperspectral imagery derived from UAS to classify different riparian plant species.

4.1.4. Bathymetry, Water Surface Elevation Survey. Bathymetry of bodies of water is required to characterize river morphology and monitor river restoration. Shallow rivers can be surveyed using methods such as total stations or RTK-GNSS, which offer high accuracy, but they are limited on a point basis, and surveying becomes difficult as the survey area or river depth increases. UAS can provide much broader and more homogeneous and contiguous spatial coverage. UAS-based bathymetry surveys are more likely to be conducted in river areas since the survey vessel with sonar equipment is very difficult to operate in the river current [184]. In addition, a bathymetric survey using an echo sounder is very difficult to apply in shallow coastal waters.

The Beer-Lambert rule, which defines the absorption effect when light flows through a transparent medium, is used to derive the flow depth from brightness values in images using remote sensing. Multi-spectral, panchromatic, and colour images can be used for this purpose. Shore shading and surface turbulence cause problems in bathymetry estimation [185]. Incorrect georeferencing, poor lighting conditions, and undesirable atmospheric conditions can adversely affect bathymetry derivation from optical data. The refraction effects that should be taken into account make it difficult to determine the bathymetric river area. There are multi-media photogrammetry techniques that consider compensation factors on every image perspective to reduce this inaccuracy [34]. Lejot et al. [116] used image processing techniques such as median filtering, histogram matching, and sub-grouping of the images to remove these inaccuracies. Flener et al. [176] combined UAS-based imagery (for optical bathymetric mapping) and mobile laser scanning data to create a bathymetric model of the river channel along with a digital terrain model of point bars of a meandering river. Structure from motion workflows and optical bathymetric mapping can also be coupled to create fluvial

terrain models [186]. However, many problems still affect unmanned image-based bathymetry [155]. Fixed-wing UAVs are less used for bathymetry studies than quadcopters and other UAVs. Woodget et al. [106] used UAS-based hyperspectral sensors for bathymetric mapping. Pan et al. [187] used LiDAR for this. LiDAR-based bathymetry has many challenges. Phenomena such as water volume scattering, water surface reflection and refraction, and turbidity reduce the signal-to-noise ratio (SNR), making processing more difficult. There are also spectral depth approaches for bathymetric mapping. Shintani and Fonstad [188] compared the spectral depth approach to the SfM photogrammetric approach. One of the experiments [189] concluded that the difference between blue and green bands is an optimal band combination for water colour inversion-based bathymetry. Bathymetric survey methods with aerial photos are more likely to be carried out in river areas since the survey ship with echo sounders is very difficult to operate due to the river current. But Woodget et al. [106] suggested that DEM derived from SfM photogrammetry should be used cautiously as in-process models are sensitive to slope.

4.1.5. Wetlands. Wetlands are at the heart of some of the most controversial and pressing issues of sustainable water management because of their complex interrelationships with the hydrological cycle and their critical dependence on water supplies [190]. Chabot and Bird [128] used UAVs for precise, fine-scale mapping of the water-vegetation interface. Multi-temporal water level changes can be detected using UAVs [33, 129]. Chabot et al. [191] applied object-based image analysis to UAS imagery for invasive species monitoring in a wetland. UAV-based LiDAR and hyperspectral sensors are yet to be deployed for various studies. Kalacska et al. [134] applied UAS photogrammetry to study tidal wetlands. Xia et al. [136] presented a novel method for sub-pixel-scale mapping of wetland flooding for satellite imagery using UAS imagery.

4.1.6. Soil Moisture. Although soil moisture contributes quantitatively to the overall global water balance [192], it is of considerable importance for water resource management [193]. According to Chabot and Bird [128], surface soil moisture (SSM) is a critical component of soil water balance that manages water and energy exchange at the surface/ atmosphere interface. In addition, soil moisture is a proxy for sub-surface water in the unsaturated zone above the water table.

Hassan-Esfahani et al. [194] used high-resolution UAS images (RGB, NIR, and thermal) and other derived images (normalized difference vegetation index (NDVI), enhanced vegetation index (EVI), vegetation condition index (VCI), and vegetation health index (VHI)) and stored capacity as inputs to an artificial neural network (ANN) model for estimating SSM. Such a modelling process is inherently subjective, and it is location-dependent and time-dependent. Further investigations under different conditions are needed to strengthen such models. Irrigation water management can be linked to the created soil moisture maps for planning and application rates.

Ge et al. [195] used a machine learning algorithm for spectral indices derived from UAV-based hyperspectral data estimation of soil moisture content. Acevo-Herrera et al. [196] described a radiometer system that performed soil moisture mapping from a small, low-altitude UAV platform. They obtained soil moisture with estimated absolute errors between 1% and 6% for the homogeneous agricultural plots. Another application of UAVs was the estimation of water infiltration rates using hyperspectral sensors [43].

4.1.7. Water Quality Monitoring. Water quality can be modelled from inputs of UAS-RS. These models can be roughly divided into three general classes, (a) radiative transfer models, (b) analytical models, and (c) empirical models [197]. Using remote sensing in conjunction with other water quality monitoring methods has four advantages:

- (1) Enables more efficient monitoring of spatial and temporal variations by providing a synoptic view of the entire body of water.
- (2) Provides a synchronized image of water quality across a range of lakes over a large area.
- (3) Provides a detailed historical record of water quality in a specific area and trends over time.
- (4) Helps prioritize sampling locations and survey times in the field.

When conducting water quality studies or attempting to predict water productivity using remote sensing, the subsurface volumetric radiance is the parameter used. This irradiance combines incident solar radiation and radiation reflected from the sky, which passes through the air-water interface, interacts with the water and organic and inorganic elements, and then exits the water column without reaching the ground. Water bodies can be surveyed at multiple locations, elevations, and viewpoints with a UASbased spectrum reflectance measurement system, making repeated measurements over the same site to limit the effects of spectral variations. Another benefit of UAS is that all of its sensors can be calibrated for water surfaces, allowing users to capture high-quality, low-SNR photos, which is not possible with satellite imagery, which is calibrated to measure land surface reflectance. Zeng et al. [198] discussed the testing and development of a low-cost UAS-based reflectance survey tool for retrieving water quality information. However, there are many challenges that UAS technology has to face in this area. This includes the development of appropriate atmospheric correction methods, developing better sensors, and developing new and existing algorithms to derive water quality information from raw sensor data. When mapping the trophic status of small reservoirs, UAS offers better value for money [97]. Freshwater water quality is highly dependent on the condition of aquatic vegetation. Therefore, when assessing

water quality, many researchers look to ecological ratings. Flynn and Chapra [122] provided a passive method for remote sensing of submerged aquatic vegetation in a shallow river using UAVs. Su et al. [97] applied a UAVmounted multi-spectral sensor to monitor water quality in small reservoirs. In this context, the pixel-by-pixel matching algorithm (MPP) has to be mentioned and has been used in many UAV-based water quality monitoring studies [97, 197, 199]. Hyperspectral sensors were used to retrieve suspended matter concentration (SSC) [41]. The study used the least squares support vector machine model. Hyperspectral data have also been used to derive water quality parameters [41, 42, 135]. Initially, these studies use ML algorithms such as SVM and ANN. Zhang et al. [135] applied the deep learning model to retrieve water quality parameters.

Water sampling has become a key activity in the management of freshwater resources. UASs are used not only for remote sensing purposes but also for sampling purposes to monitor water quality. A study [200] showed that the UAS mechanism could collect samples similar to manually collected samples. These UAS significantly reduces the effort and time required for sampling. Ore et al. [201] developed an autonomous UAS-based water sampler. Doi et al. [99] used UAVs to extract environmental DNA (eDNA) from a reservoir. Similarly, Terada [102] sampled a crater lake in Japan. UAVs have been developed that can overcome the speed limitations of flowing waters [202]. The UAVs used for water sampling should be able to support the additional weight of the sampling mechanism, the water collected, and additional provisions for an emergency landing on the water surface. From the literature, many examples of such UAVs, capable of navigating in both air and water, can be used for water sampling [203, 204]. Song et al. [100] discussed the advantages and limitations of UAV-based sampling along with the manual and sensorbased methods in limnology. Ore and Detweiler [205] presented a UAS-based system that includes a submersible sensor probe that can be used to measure water properties that can keep the target submerged. Benson et al. [206] developed a sampling system called DOWSE, DrOne Water Sampling SystEm. Their goal was to study the spatial distribution of microorganisms in freshwater lakes. There is now a need for future drone-based water sampling studies to adapt more robust statistical experimental designs to examine the variability and precision of the collected data 20].

Chung et al. [96] extended the concept of water sampling by UAS to the temperature measurement of the water column (Figure 5). They proposed and tested an automated temperature sensing system based on using an unmanned air system to quickly obtain 3D thermal maps of bodies of water by lowering a temperature probe into the water at controlled depths with an unmanned air system. Demario et al. [207] also developed another UAS-based water temperature measurement system consisting of an IR camera and an immersible temperature probe. Table 10 gives the insight into different water quality parameters derived from different UAS-based sensors.



FIGURE 5: Image of UAS-based temperature sampler [96].

Water property	Response parameter	Remote sensing indicator	Sensor used	Remark
Hydrology	Water level	Bathymetry	Multi- spectral	Brightness levels in imagery correspond to the depth
Temperature	Temperature	Surface temperature	Thermal	Spatial patterns can predict algae blooms
Transparency	Turbidity and Secchi depth	Secchi disc depth	Multi- spectral	
	Algal growth	Chlorophyll-a	Multi- spectral	NDVI can be used. Spectral mixing could be a troublemaker.
Biota	Phenology	Time series analysis	Multi- spectral	
	Species analysis		Multi- spectral	
Temperature anomaly	Temperature	Surface temperature	Thermal	Locate groundwater discharge to surface water
Thermal pollution from industrial sources	Temperature	Surface temperature	Thermal	

TABLE 10: Water quality parameters as derived from	UAS	Ss
----------------------------------------------------	-----	----

4.1.8. Water Pollution and Wastewater. In recent years, UAV remote sensing is gradually being used to monitor and assess pollution of the aquatic environment. The main application of UAVs for water pollution is the monitoring and mapping of the algal bloom population. Recently, many studies have used UAS-based optical sensors to monitor algal blooms in surface waters, demonstrating their ability to quantify algal species using a variety of indices such as the Normalized Difference Vegetation Index (NDVI), Algal Bloom Detection Index (ABDI), and Green Leaf Index (GLI). However, there are two main problems.

- (i) The sub-grid heterogeneity can complicate imagery data assimilation.
- (ii) The mismatch between model and measurement scales in the vertical direction represents another problem that needs to be investigated for the assimilation of the remotely sensed data into water quality models.

Another aspect of dealing with water pollution is water spillage. UAVs have been deployed to inspect spills of contaminants and to estimate spill volume and area to quantify pollution [152, 208, 209, 209]. UAS has been mainly used for oil spill mapping and monitoring [49, 152, 210–213]. Coal ash spills have also been assessed and monitored using UAS [208]. Most approaches applied computer vision approaches to oil spill detection. Oil spill monitoring application has seen the application of UAS-Swarm. Kaviri et al. [213] designed a multi-UAS control framework for oil spill mitigation. A novel spray adjustment strategy was also proposed in this study to combat oil spills.

So far, there have only been a few applications of drones, especially in wastewater treatment. Sancho Martínez et al. [214] developed a UAS-based image-based methodology for autonomous inspection of trickling filters and activated sludge systems. Wong et al. [215] used UAVs to inspect the foam in the covered anaerobic lagoon of a sewage treatment plant. Burgués et al. [216] used UAS-based gas sensors for real-time monitoring of emissions from wastewater treatment plants.

UAS-based infrared thermography, together with modern data processing and visualization tools, can aid in the study of numerous environmental issues, including pinpointing pollution point sources and determining the best path between sources and targets. Compared to conventional pollution-source detection methods, UAS-thermography makes finding unlawful sanitary sewers and storm-drain connections, illicit discharges, and other reasons for the surface water contaminationvery easy [159]. For studying such contamination, night flights have been recommended in the literature [217].

Tracer studies can also be performed using UAV data, which can help study water pollution [76]. UAV-based remote sensing can compensate for other platforms' low spatial and temporal resolution for such river investigations. Geraeds et al. [218] showed that UAVs can be used to monitor the spatiotemporal distribution of plastic waste in rivers. Plastic pollution of the marine environment has increased dramatically in recent decades and poses a severe environmental threat to numerous settings worldwide. UAS remote sensing can be used for (i) detection, (ii) identification and categorization, (iii) quantification, and (iv) mapping and estimating the accumulation rate of marine debris. The process with UAVs is faster [219] and more reliable. Ferrara et al. [220] demonstrated the use of remote sensing-based surveillance of coastal waters using thermal imaging to study the extent of pollution. This study hierarchically used satellite, helicopter, and drone data to monitor water quality in pollution scenarios. Consumergrade UAVs can be used to detect these pollutants [221], particularly when used in conjunction with machine learning [222] and deep learning techniques [144, 223-226]. However, these automated approaches still need to be improved [222]. Gonçalves et al. [222] suggested that manual image screening of the orthophoto should be preferred when more careful characterization of marine litter is required.

4.1.9. Coastal Water Management and Oceans. Drones find tremendous utility in monitoring and mapping oceans and coastal areas [153, 227, 228]. Regular monitoring in coastal

areas to assess topographic changes requires very detailed and fast mapping technologies that UAS can provide. Another advantage of UAS technology is its flexibility, which allows UAV surveys to be timed to avoid flooding, insufficient light conditions, etc. [134]. Since light is refracted on the water surface, the most difficult problem of UAVassisted coastal photogrammetry is to incorporate and exploit surface and underwater imagery records from overhead images [229]. Another difficulty is calibrating UAS-derived elevation information to local sea levels [230]. One of the problems for UAS surveys in coastal environments is the extensive area of salt marshes, tidal flats, etc., where it is challenging to establish ground control points [23]. Jaud et al. [231] suggested that artificial georeferenced targets can be used, which should be easily visible in different beach lighting conditions.

UAS can be used for coastal zone mapping [232], coastal flooding monitoring [142, 149], river deltas [151], tidal reefs [148], dynamic tidal inlets [8, 233], sea-level rise [230], surveying of coastal structures [234], sea-level rise scenario simulation [142], analysis of shoreline changes [9], and beach sediment changes [151]. The drone survey was used to validate satellite monitoring of the Yellow Sea green tide disaster [235]. Similarly, a study [150] to estimate the algal biomass in the Yellow Sea was conducted using UAV data and satellite imagery.

4.1.10. Urban Storm Water. McDonald [90] reviewed the application of urban stormwater drones. UAVs can be used to assess damage from urban flooding [13]. Many studies have shown this applicability of UAS [236]. The study of stormwater contamination is where UAV-based thermal imaging can be used [217].

4.1.11. Cryosphere. The cryosphere is a critical water resource, and snow cover extent and depth are important parameters affecting energy and water balance. Frozen water occurs in remote areas that are difficult to access, especially mountainous areas, and there are limited resources for ground measurements. Unmanned aerial systems can prove to be a boon. The main applications in the cryosphere include DSM generation [154], change detection, snow depth estimation [65, 237, 238], tundra vegetation mapping [239, 240], ice-wedge polygon mapping [241], and so on.

Gaffey and Bhardwaj and Bhardwaj et al. [18, 89] gave a detailed report on the applications of UAVs as a remote sensing platform in cryospheric studies. These reviews showed that modern UAVs have all the necessary equipment and features that can make them useful for glaciological research. But what is needed is an improvement. Since the cryospheric application of UAVs has already been covered in detail, the details will not be covered in this review.

In their study, Ramsankaran et al. [140] summarized various challenges for surveying glaciers and found that the choice of UAV launch/landing sites affects the survey and recommended selection criteria for the most suitable launch/landing sites. Snow-covered surfaces pose photogrammetric challenges as there is a lack of contrast and the surface

reflectance is very high. Therefore, weathered old snow under cryospheric conditions is better suited for SfM processing but still not reliable for repeated surveys [18].

4.2. Sub-Surface Water Resources. Groundwater is difficult to map from UAVs because the target of interest is not directly observable from the air. A gravity survey using an unmanned aerial vehicle (UAV) can help determine alluvial groundwater storage and specific yield, which is an important parameter for the long-term management of groundwater resources. Dedicated UAV-based gravimeters are being developed [67-69] that can measure the spatial and temporal variation of sub-surface density with a resolution and accuracy which is almost comparable to terrestrial gravimetry. Advances in gravimetry combined with the development of UAV-based gravimeters will make it possible to analyze changes in groundwater over time. In the future, UAV gravity surveys can be successfully used to determine the specific yield and groundwater storage of alluvium, as has been the case with ground-based time-lapse gravity surveys [242]. This area of UAV-based gravimetric analysis of groundwater needs to be explored.

Under certain conditions, UAS remote sensing can also infer groundwater by quantifying temperature or electrical conductivity anomalies [243]. One such study [19] used the availability of visible surface water as an indication of groundwater and used UAS surveying to model the water table.

UAS data inputs are instrumental in groundwater modelling. Groundwater models have become the most sophisticated tool for decision makers in groundwater management. There are two main ways in which UAS-based remote sensing data can be applied: (1) using UAS-acquired data to construct distributed sets of input parameters for a model and (2) providing constraints to models during the calibration of models by acquired data FH. UAV-based geophysical surveys allow the identification of faults and dykes and mapping of lithology and its alterations and depth of magnetic features (e.g., [244-246]). These data help to develop more realistic aquifer models. The upper limit of an aquifer is also the surface of the topography that restricts the water table. Surface elevations can be determined with subdecimeter accuracy using UAV imagery processed with structure from motion algorithms. This UAS technology produces DEMs that rival LiDAR in terms of accuracy but is much cheaper.

4.3. Irrigation and Water Structure Monitoring. Irrigation control is related to sustainable water management. UAV-based remote sensing can be used for real-time (daily or weekly) monitoring of various parameters from the field. These may include the following, which can help with irrigation planning and water management:

- (i) Water-related information: this includes its quantity and quality-related aspects with spatiotemporal dimensions.
- (ii) Soil-moisture related information.

(iii) Vegetation-related information: vegetation index includes its quantity and quality-related aspect with temporal and spatial dimensions, such as crop phenology.

Monitoring of structures related to irrigation or revetments is necessary to ensure their longevity and thus contributes to sustainable water management.

Rathinam et al. [119] proposed a real-time image-based detection algorithm for autonomous inspection of various linear features (canals, rivers, and pipelines) using a UAV. But that was a preliminary concept. Chao et al. [247] provided an overview of UAV-based irrigation control and water management system (hardware and software).

Perea-Moreno et al. [248] used object-based image analysis for the automated classification of UAV videos. This strategy used the hierarchical temporal memory (HTM) learning algorithm. Multi-spectral information from UAS was used to develop a decision support system to regulate irrigation rates [7]. UAVs can also be effective in weed assessment and hydraulic efficiency [249].

Rathinam et al. [119] presented a study demonstrating a structure recognition algorithm that can identify and localize a channel. This type of monitoring is critical to ensure the reliability and life expectancy of canals and other irrigation structures. Irrigation channels were reconstructed virtually with UAS by Brinkhoff and Hornbuckle [250] and monitored for the presence of aquatic weeds. Kadapala et al. [251] implemented UAV photogrammetry to estimate the capacity of an irrigation tank. The UAV data were used to generate an EAC (elevation-area-capacity) curve for the irrigation tank. The use of drones is only successful for small tanks and reservoirs.

Miller et al. [60] used NASA's UAVSAR L-band data for successfully monitoring the California Aqueduct and showed the advantage of UAVSAR over satellite SAR data. Leaks in river levees can be detected with UAS-based thermal levees [252]. Kubota et al. [173] proposed a river facility maintenance management system based on 3D point clouds derived from UAS photogrammetry. Chen et al. [253] used UAS-derived dense point clouds to inspect revetments along urban rivers.

Monitoring complex-shaped dams by UAV surveys is a complex process that has been reinforced by other survey techniques (TLS, GPS, and total station). Several studies on the use of drones for dam inspections have been conducted over the last decades [254–256]. These surveys can be performed reliably and efficiently when the impact of the number and location of GCPs on model accuracy is known a priori [254]. The markers' position affects the survey's accuracy and needs further research [255]. UAVs have found their application in the design of terrace drainage networks. Pijl et al. [257] used the topographical data derived from the UAV survey to analyze and design the drainage networks in terraces in Italy.

Tile drainage systems remove excess water from fields, benefiting both the environment and the economy. Farmers and natural resource managers can better mitigate any adverse environmental and economic impacts by monitoring tile operations. UAS thermal data can provide additional insight into field tile mapping by assessing temperature changes within a field.

5. Software

There are a lot of commercial and a few open-source software programs that allow photogrammetric processing of UAV images. Some of these are summarized in Table 11.

6. Issues

UAVs in water resource management and monitoring open new avenues for research and development. This versatile platform has enormous potential. However, it still faces many problems and bottlenecks in its application in this field. In our literature review, we found the following challenges:

- (1) The size of the problem and the mode of application: most water resource problems are extensive in space and time but can be severely limited in the time of their occurrence, especially in the case of catastrophic events and disasters. Given UAVs' intrinsic flight time limitation, their use in large-scale transport challenges should be planned and implemented rigorously. A performance like this would require either UAVs with advanced technology or a swarm of UAVs to increase their capabilities. The biggest shortcomings of unmanned aerial systems are the limited flight duration, the weather, and the regulatory challenges. UAVs must be used responsibly. The result of a UAV survey often reflects human error. The forward and side overlap is one of the user-controlled properties that affect the quality of the orthomosaic. For mapping surveys that require high precision, the recommended forward and side overlap value is at least 80% [258]. Researchers and practitioners in the field of WRMM need to interact with professionals in the field of UAS technology to discover appropriate existing solutions or to develop new technologies to solve specific WRMM problems. UASs face particular problems when inspecting the natural and man-made aquatic environment. Therefore, knowing the strengths and limitations of UAV technology is crucial for selection, development, and mission planning.
- (2) Flight time: the flight time of UAVs significantly impacts their application. It limits the area to be captured at once. So, if a larger waterbody or study area has to be monitored, multiple flights or multiple UAS are required. The flight time of UAVs varies from platform to platform.
- (3) Payload capacity: commercially viable UAVs are generally smaller in size and therefore cannot carry much larger payloads. The payload capacity dictates the type of sensors that can be utilized. Their payload mainly constrains UAVs. Their payload capacity

mainly constrains UAVs. Thus, this also limits the research applications.

- (4) Reliability: weather conditions again limit the use of UAS. The UASs are vulnerable to wind and rain. Many terrain conditions also require specially designed UASs for their surveillance. UAV platforms are prone to instability when wind speeds increase. Because lighting conditions vary between images and flights, cloudy skies can cause image quality issues. Another factor is the availability of GPS for UAS operations [2].
- (5) Data interoperability with other Earth observation platforms is also an issue. There is heterogeneity in data collection from different platforms and sensors. The synergistic use of UAVs, unmanned surface vehicles, and unmanned submerged vehicles has to be explored and researched from the WRMM perspective. The UAV/satellite synergy potential is still underexploited [3].
- (6) Legal issues and drone security: in many ways, operating unmanned aerial vehicles (UAVs) safely and efficiently over water resources and infrastructure is daunting. When surveillance needs to be done on a large scale or in challenging terrain, drone security becomes an issue.
- (7) Different data acquisition conditions for different applications: in the field of mapping and monitoring water resources, UAVs need to make some considerations specific to working over water. In the production of remote sensing data from visible band airborne sensors, distortions resulting from the reflection of light from water-based surfaces have long been a problem. The UAS operator must be aware of such difficulties in any data collecting situation over water surfaces because they can appear in complex ways in good-looking data [259]. Different WRMM problems have different requirements, such as requirements for spatial resolution, temporal frequency, flight path, and hardware (payload) requirements.
- (8) Lack of geospatial standards and protocols: there is a lack of standards and best practices regarding UAS data collection, planning, data processing, accuracy assessment, feature extraction, etc., especially with regard to application in WRMM [172]. The variation in these methods introduces uncertainties that represent a bottleneck in the widespread use of UAS data. There is a need to establish a standardized procedure for data collection, processing, and output [260]. Few researchers have attempted to recommend practices [261] and develop protocols [262], but this has been limited to the marine environment. Much needs to be done.
- (9) Software and hardware challenges: the data collected by the UAV and its secondary products can be large and require huge digital storage. Photogrammetric processing and AI-based processes are

Advances in Civil Engineering

Softwara nama	Davalopar/propriator	Link/license type	Capability
Software name	Developer/proprietor	Link/license type	Navily upgraded version is called acted
Agisoft photoscan	Agisoft	Commercial	metashape
MATLAB	Mathworks	Commercial (student license available)	Performs geodetic calculations. In a single profile, combines vector and raster datasets Terrain and elevation analysis
OpenDroneMap	OpenDroneMap	Open-source	Transforms 2D pictures into: Point clouds that have been classified. Textured 3D models Imagery that has been georeferenced and orthorectified. Digital elevation models with
Pix4D mapper	Pix4D	Commercial	Classification of the point cloud automatically Identification of digital surfaces that are flat and smooth Measures surface, distance, and volume
QGIS	QGIS community	Open-source	Vector analysis, raster analysis, sampling, geoprocessing, geometry, database management. Composes maps Analyzes data
OpenCV	OpenCV	Open-source	Reads and writes pictures. Records and saves videos. Processes images. Performs feature detection. Detects specific objects in videos or images. Estimates movement and track objects.
Drone2Map	ArcGIS	Commercial (organizational license if available)	ArcGIS Drone2Map is a desktop application that converts drone still images into useful information products in ArcGIS Rapid processing
Menci software	Menci software Srl	Commercial	Creation of 3D aerial cartographic inspection and plotting DEM editor toolsets for volume, profile, and advanced DSM/DTM analysis
Autodesk ReCap	Autodesk	Commercial (student license is available)	Viewing projects in 3D, annotating and sharing data to other software Physical-world detail transformation into digital assets
Maps made easy	Automotive Data Research (ADR)	Commercial (free for small executions)	Orthophoto map and 3D model generation, 3D model-based stitching, stockpile volume measurement
3DF zephyr	3D flow	Open-source	3D reconstruction and scanning, reconstructing 3D models from pictures Generates realistic orthophotos, DTM and DSM models, statistics, and project reports
PrecisionHawk 3D map software	PrecisionHawk	Open-source (mobile application)	Rates the image quality. Adds ground control points to ensure data accuracy. Compresses data for cloud upload.
Correlator3D™	SimActive	Commercial	Aerial triangulation (AT) produces DSM, DTM, ortho-mosaics, 3D models, and vectorized 3D features
Drone deploy 3D	Drone deploy	Commercial (free 14-day trial for mobile phone application)	In-field insights, data analysis, and virtual walkthroughs. Provides reports.
Drone photogrammetry software	Propeller	Commercial	Aerial images to 3D models, accurate measurements
Bentley context capture	Bentley	Commercial	Generating multi-resolution 3D models at any scale, producing reality meshes, orthophotos, DSM, and point clouds
ESRI sitescan	ESRI	Commercial	Drone flight planning, fleet management, image processing, image analysis, 3D textured meshes

TABLE 11: UAV photogrammetry data processing software.

Software name	Developer/proprietor	Link/license type	Capability
Agisoft metashape	Agisoft	Commercial	Photogrammetric triangulation, dense point cloud, DSM/DTM generation Georeferenced ortho-mosaic generation
Regard3D		Open-source	Reconstructing 3D models from pictures
COLMAP	Johannes L. Schönberger, jan-michael frahm, and marc pollefeys	Permissive free software	Reconstructing 3D models from pictures
MicMac	IGN (French national geographic institute) and ENSG (French national school for geographic sciences)	Open-source	Reconstructing 3D models from pictures
SOCET GXP	BAE systems	Commercial	Orthomosaic, surface terrain model, bare Earth terrain model, 3D point cloud
Trimble inpho UASMaster	Trimble inc	Commercial	Detailed 3D models and point clouds
ELCOVISION 10	PMS AG Switzerland	Commercial	Detailed 3D models and point clouds

TABLE 11: Continued.

computationally intensive. They require systems with high processing power. Methods of data processing and analysis that effectively exploit these data's high spatial and temporal frequency need to be further explored. Apart from that, there are also many challenges at the data level. Image alignment and radiometric accuracy of thermal images captured by UAS and inexpensive TIR cameras face several challenges.

7. Future

7.1. Diversity in UAVs as a Platform. New platforms are being developed that are more compact, lighter, cheaper, safer, and more reliable. Sensors are also being miniaturized. In the future, the focus will be on increasing the energysaving capabilities of UAS. With the development of a wide variety of sensors in future studies, it must be emphasized whether the uncalibrated UAV cameras and sensors can be robust measurement tools for applications in WRMM.

7.2. Development of New Methods. UASs face particular problems when inspecting the natural and man-made aquatic environment; therefore, knowing the strengths and limitations of UAV technology is crucial for selection, development, and mission planning. New analysis methods for decision making in water resource systems need to be developed that consider data collection as a goal in the decision-making process. Such methods would explicitly consider the cost and value of data collection, allowing more data to be collected to reduce uncertainty at the expense of decision speed or the economic cost of data collection. Traditional data processing is maturing. Recent developments in the field of artificial intelligence are used in other disciplines. The field of water resource management and mapping can benefit from the application of artificial intelligence. UAS-collected data have a high resolution, so the data volume is large in many cases, which poses a challenge for the analyst. The AI algorithms offer a solution to this problem. We encourage the water resources community to implement the developments for WRMM. UAS applications

in the field of WRMM will require new modelling, computational, and mathematical approaches to assimilate data collected by traditional remote sensing and UAS to help identify optimal and timely decisions. The future of using UAS for WRMM will be dominated by advanced algorithms and predictive tools with a focus on analytics-based data mining of crucial information. Ongoing developments in AI, ML, and DL are likely to improve the efficiency and scalability of UAV analysis approaches in WRMM. Virtual and augmented reality can also be helpful in cooperation with UAS exits. Due to its immersive nature, the combination of deep learning with virtual and augmented reality environments represents an important research topic for the effective study of complex environmental phenomena that are difficult to organize in reality.

7.3. The Cost Will Come Down. With advances in electronics and materials science, the cost of UAS is predicted to decrease. This will increase both the popularity and range of drones. In combination with open-source initiatives, this will also promote citizen science. Citizen science is already gaining popularity in the field of mapping and monitoring water resources [263]. Citizens can support the mapping and monitoring of water resources at a low cost and contribute to the data pool in data-poor and understudied areas [264]. Crowdsourced data could be an important complementary data source for monitoring water resources.

7.4. Real-Time Usage. With the demand for a higher degree of automation and a reduction in the time between collection and output, real-time onboard data processing is increasing daily.

7.5. Cost-Benefit Analysis of UAV samplers. When it comes to the cost-benefit analysis of water samples, the cost-benefit analysis is still unclear [20]. These cost-benefit analysis studies may also include health and safety and biosecurity risks. UAV-based water sampling should be compared to other sampling methods helpful in making informed decisions about sampling methods (based on careful cost estimates).

7.6. Synergies with Other (Satellite) Datasets. The UAV/ satellite synergy potential is still underexploited. Data fusion can be studied for different sensors for different applications, similar to those in other fields [234]. In the future, there is a lot of scope for developing the integration of satellite intelligence pipelines with drone-based knowledge [265]. The realm of UAV/satellite synergy potential remains untapped for many areas, particularly water resource mapping and monitoring [3]. UAVs can bridge the gap between in situ observations and satellite data [153]. There is cause for optimism that UAS and data collected by different payloads can be used at different scales in monitoring and mapping water resources. However, this optimism must be balanced with a dose of reality regarding technological and legal challenges. Many new techniques for UAV-derived data such as sensor data fusion need to be explored. Addressing these challenges to extend proof-of-concept studies to inflection points and realize the hidden potential of UAS technology requires broader, collaborative methodologies supported by strong funding initiatives. New analytical methods for mapping and monitoring water resource systems need to be developed. Inexpensive UAV sensors may pose problems in synergizing with satellite data [3, 266]. Another aspect to study is the combination of UAVs with aquatic drones (USVs) and unmanned surface boats [10]. Data input from a combined use will improve mapping compared to UAV alone [267]. Few studies have attempted to investigate this in the past [268]. But this aspect has not yet been fully explored.

7.7. Development of Swarm Intelligence. Another area that is increasingly being researched is swarm intelligence and swarm intelligence. The swarm can be applied in many innovative and diverse ways in the field of water resource mapping and monitoring. There have been few studies [211, 247, 269] that attempted to explore the applicability of swarming to different aspects of WRMM. Nevertheless, swarm development itself faces many challenges [267]. Long-term WRMM swarm systems have challenges in terms of scalability, maintainability, safety, and flight endurance. Studies have been limited to testing algorithms or data processing. This swarm system needs to be explored more closely for WRMM.

7.8. Deregulation. Various studies continuously demonstrate that UAS technology can be safely used in mapping and monitoring water resources. This also leads to cautious deregulation of the technology and the removal of legal hurdles by the administrative authorities [270]. All this will lead to increased use of UAS systems in the field of WRMM.

8. Conclusion

The authors attempted to provide an overview of various sensors and the application of unmanned aerial systems in water resource management and monitoring. A wide range of sensors available can cater to WRM practitioners' needs. There are several helpful recommendations in the literature

that have been compiled to offer direction for using drones for water resource applications. The information examined was primarily practical and applicable to a variety of subsectors, including river mapping, bathymetric mapping, water quality, wastewater, and coastal mapping. The field of water resources is vast; with each coming day, researchers are inventing new ways to use UAS for this cause. To support WRM applications, UAS should be applied in a multi-disciplinary manner, including different approaches and topics. The UAS technology has many advantages such as low cost, high spatial resolution, operator-subjective temporal resolution, and non-intrusive methodology in many WRMM applications. The field of UAS application in WRMM is yet to be matured and still faces many challenges such as battery life, data interoperability, legal hurdles, and so on. These challenges and issues have been compiled in this research. The authors have identified the areas in which future research can take place. The use of these unmanned aerial systems for water resource mapping, monitoring, and hydrological research remains experimental in many places. This study is very important, especially for new researchers. The information is easily adaptable to different areas of WRMM.

Data Availability

No data were used to support this study.

Disclosure

The sponsors had no role in the design of the study or in the collection, analysis, or interpretation of data when writing the manuscript or when deciding to publish the results.

Conflicts of Interest

The authors declare that they have no conflicts of interest.

Acknowledgments

The authors would like to thank the Indian Institute of Technology Roorkee for using their resources. The authors are very thankful to Prof. Laszlo Bertalan (University of Debrecen, Hungary) and Mr. Prashant Singh (Indian Institute of Technology Roorkee, India) for their suggestions.

References

- Z. W. Kundzewicz, "Water resources for sustainable development," *Hydrological Sciences Journal*, vol. 42, no. 4, pp. 467–480, 2009.
- [2] M. V. Nicolas, S. L. García, L. Barbero, V. O. Ruiz, and Á. S. Bellon, "Applications of unmanned aerial systems (UASs) in hydrology: a review," *Remote Sensing*, vol. 13, no. 7, p. 1359, 2021.
- [3] E. Alvarez-Vanhard, T. Corpetti, T. Houet, A.-V. Emilien, and H. Thomas, "UAV & satellite synergies for optical remote sensing applications: a literature review," *Science of Remote Sensing*, vol. 3, Article ID 100019, 2021.

- [4] K. Jain and A. Pandey, "Calibration of satellite imagery with multispectral UAV imagery," J. Indian Soc. Remote Sens, vol. 49, no. 3, pp. 479–490, 2021.
- [5] G. Pajares, "Overview and current status of remote sensing applications based on unmanned aerial vehicles (UAVs)," *Photogrammetric Engineering & Remote Sensing*, vol. 81, no. 4, pp. 281–330, 2015.
- [6] S. Manfreda, M. McCabe, P. Miller et al., "On the use of unmanned aerial systems for environmental monitoring," *Remote Sensing*, vol. 10, no. 4, p. 641, 2018.
- [7] X. Shi, W. Han, T. Zhao, and J. Tang, "Decision support system for variable rate irrigation based on UAV multispectral remote sensing," *Sensors*, vol. 19, no. 13, p. 2880, 2019.
- [8] N. Mohamad, M. F. Abdul Khanan, A. Ahmad, A. H. Md Din, and H. Shahabi, "Evaluating water level changes at different tidal phases using UAV photogrammetry and GNSS vertical data," *Sensors*, vol. 19, no. 17, p. 3778, 2019.
- [9] C. D. Troy, Y. T. Cheng, Y. C. Lin, and A. Habib, "Rapid lake Michigan shoreline changes revealed by UAV LiDAR surveys," *Coastal Engineering*, vol. 170, Article ID 104008, 2021.
- [10] P. Duan, M. Wang, Y. Lei, and J. Li, "Research on estimating water storage of small lake based on unmanned aerial vehicle 3D model," *Water Resources*, vol. 48, no. 5, pp. 690–700, Sep. 2021.
- [11] E. Husson, O. Hagner, and F. Ecke, "Unmanned aircraft systems help to map aquatic vegetation," *Applied Vegetation Science*, vol. 17, no. 3, pp. 567–577, 2014.
- [12] M. Rusnák, T. Goga, L. Michaleje et al., "Remote sensing of riparian ecosystems," *Remote Sensing*, vol. 14, p. 2645, 2022.
- [13] S. I. Jiménez-Jiménez, W. O. Bustamante, R. E. O. Capurata, and M. D. J. Pablo, "Rapid urban flood damage assessment using high resolution remote sensing data and an objectbased approach," *Geomatics, Natural Hazards and Risk*, vol. 11, no. 1, pp. 906–927, 2020.
- [14] A. Gebrehiwot, L. B. Hashemi, G. Thompson, P. Kordjamshidi, and T. E. Langan, "Deep convolutional neural network for flood extent mapping using unmanned aerial vehicles data," *Sensors*, vol. 19, no. 7, p. 1486, 2019.
- [15] L. B. Hashemi, J. Jones, G. Thompson, C. Johnson, and A. Gebrehiwot, "Challenges and opportunities for UAVbased digital elevation model generation for flood-risk management: a case of princeville, North Carolina," *Sensors*, vol. 18, no. 11, p. 3843, 2018.
- [16] A. Annis, F. Nardi, A. Petroselli et al., "UAV-DEMs for small-scale flood hazard mapping," *Water*, vol. 12, no. 6, p. 1717, 2020.
- [17] L. Hashemi-Beni and A. A. Gebrehiwot, "Flood extent mapping: an integrated method using deep learning and region growing using UAV optical data," *Ieee Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, vol. 14, pp. 2127–2135, 2021.
- [18] C. Gaffey and A. Bhardwaj, "Applications of unmanned aerial vehicles in cryosphere: latest advances and prospects," *Remote Sensing*, vol. 12, no. 6, p. 948, 2020.
- [19] M. M. Rahman, G. J. McDermid, M. Strack, and J. Lovitt, "A new method to map groundwater table in peatlands using unmanned aerial vehicles," *Remote Sensing*, vol. 9, no. 10, p. 1057, 2017.
- [20] H. T. Lally, I. O'Connor, O. P. Jensen, and C. T. Graham, "Can drones be used to conduct water sampling in aquatic environments?" *Science of the Total Environment*, vol. 670, pp. 569–575, 2019.

- [21] L. Huylenbroeck, M. Laslier, S. Dufour, B. Georges, P. Lejeune, and A. Michez, "Using remote sensing to characterize riparian vegetation: a review of available tools and perspectives for managers," *Journal of Environmental Management*, vol. 267, Article ID 110652, 2020.
- [22] B. M. P. Chandler, H. Lovell, C. M. Boston et al., "Glacial geomorphological mapping: a review of approaches and frameworks for best practice," *Earth-Science Reviews*, vol. 185, pp. 806–846, 2018.
- [23] R. Adade, A. M. Aibinu, B. Ekumah, and J. Asaana, "Unmanned Aerial Vehicle (UAV) applications in coastal zone management—a review," *Environmental Monitoring and Assessment*, vol. 193, no. 3, p. 154, 2021.
- [24] C. Kislik, I. Dronova, and M. Kelly, "UAVs in support of algal bloom research: a review of current applications and future opportunities," *Drones*, vol. 2, no. 4, p. 35, 2018.
- [25] L. Debell, K. Anderson, R. E. Brazier, N. King, and L. Jones, "Water resource management at catchment scales using lightweight uavs: current capabilities and future perspectives," *Journal of Unmanned Vehicle Systems*, vol. 4, no. 1, pp. 7–30, 2015.
- [26] J. L. Molina, P. Rodríguez-Gonzálvez, M. C. Molina, D. González-Aguilera, and F. Espejo, "Geomatic methods at the service of water resources modelling," *Journal of Hydrology*, vol. 509, pp. 150–162, 2014.
- [27] T. Adão, J. Hruska, L. Padua et al., "Hyperspectral imaging: a review on UAV-based sensors, data processing and applications for agriculture and forestry," *Remote Sensing*, vol. 9, no. 11, p. 1110, 2017.
- [28] F. Remondino, L. Barazzetti, F. Nex, M. Scaioni, and D. Sarazzi, "Uav photogrammetry for mapping and 3D modeling – current status and future perspectives," *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. 38, pp. 25–31, 2012.
- [29] V. Chamola, P. Kotesh, A. Agarwal, N. Gupta, and M. Guizani, "A comprehensive review of unmanned aerial vehicle attacks and neutralization techniques," *Ad Hoc Networks*, vol. 111, Article ID 102324, 2021.
- [30] K. Parvar, A. Braun, D. M. Layton, and M. Burns, "Uav magnetometry for chromite exploration in the samail ophiolite sequence, Oman," *Journal of Unmanned Vehicle Systems*, vol. 6, no. 1, pp. 57–69, 2018.
- [31] L. Debell, K. Anderson, R. E. Brazier, N. King, and L. Jones, "Water resource management at catchment scales using lightweight uavs: current capabilities and future perspectives," *Journal of Unmanned Vehicle Systems*, vol. 4, no. 1, pp. 7–30, 2016.
- [32] M. H. Gholizadeh, A. M. Melesse, and L. Reddi, "A comprehensive review on water quality parameters estimation using remote sensing techniques," *Sensors*, vol. 16, no. 8, p. 1298, 2016.
- [33] M. Kohv, E. Sepp, and L. Vammus, "Assessing multitemporal water-level changes with uav-based photogrammetry," *Photogrammetric Record*, vol. 32, no. 160, pp. 424–442, 2017.
- [34] A. Eltner, L. Bertalan, J. Grundmann, M. T. Perks, and E. Lotsari, "Hydro-morphological mapping of river reaches using videos captured with UAS," *Earth Surface Processes* and Landforms, vol. 46, no. 14, pp. 2773–2787, 2021.
- [35] L. Bertalan, I. Holb, A. Pataki et al., "UAV-based multispectral and thermal cameras to predict soil water content – a machine learning approach," *Computers and Electronics in Agriculture*, vol. 200, Article ID 107262, 2022.

- [36] H. Saari, V. V. Aallos, A. Akujarvi et al., "Novel miniaturized hyperspectral sensor for UAV and space applications," *SPIE Proceedings*, vol. 7474, 2009.
- [37] R. Hruska, J. Mitchell, M. Anderson, and N. F. Glenn, "Radiometric and geometric analysis of hyperspectral imagery acquired from an unmanned aerial vehicle," *Remote Sensing*, vol. 4, no. 9, pp. 2736–2752, 2012.
- [38] S. Natesan, C. Armenakis, G. Benari, and R. Lee, "Use of UAV-borne spectrometer for land cover classification," *Drones*, vol. 2, no. 2, p. 16, 2018.
- [39] H. Aasen, E. Honkavaara, A. Lucieer, and P. Zarco-Tejada, "Quantitative remote sensing at ultra-high resolution with UAV spectroscopy: a review of sensor technology, measurement procedures, and data correction workflows," *Remote Sensing*, vol. 10, no. 7, p. 1091, 2018.
- [40] G. Szabó, L. Bertalan, N. Barkóczi, Z. Kovács, P. Burai, and C. Lénárt, "Zooming on aerial survey," *small fly. Drones Appl. Geogr. Obs*, Springer, Berlin, Germany, pp. 91–126, 2017.
- [41] L. Wei, C. Huang, Y. Zhong, Z. Wang, X. Hu, and L. Lin, "Inland waters suspended solids concentration retrieval based on PSO-lssvm for UAV-borne hyperspectral remote sensing imagery," *Remote Sensing*, vol. 11, no. 12, p. 1455, 2019.
- [42] Y. Zhang, L. Wu, H. Ren et al., "Mapping water quality parameters in urban rivers from hyperspectral images using a new self-adapting selection of multiple artificial neural networks," *Remote Sensing*, vol. 12, no. 2, p. 336, 2020.
- [43] N. Francos, N. Romano, P. Nasta et al., "Mapping water infiltration rate using ground and UAV hyperspectral data: a case study of alento, Italy," *Remote Sensing*, vol. 13, no. 13, p. 2606, 2021.
- [44] M. G. Seo, H. S. Shin, A. Tsourdos, R. Taghizadeh-Mehrjardi, and K. Schmidt, "Soil moisture retrieval model design with multispectral and infrared images from unmanned aerial vehicles using convolutional neural network," *Agronomy*, vol. 11, no. 2, p. 398, 2021.
- [45] H. Hoffmann, H. Nieto, R. Jensen, R. Guzinski, P. Zarco-Tejada, and T. Friborg, "Estimating evaporation with thermal UAV data and two-source energy balance models," *Hydrology and Earth System Sciences*, vol. 20, no. 2, pp. 697–713, 2016.
- [46] C. Brenner, C. E. Thiem, H. D. Wizemann, M. Bernhardt, and K. Schulz, "Estimating spatially distributed turbulent heat fluxes from high-resolution thermal imagery acquired with a UAV system," *International Journal of Remote Sensing*, vol. 38, no. 8–10, pp. 3003–3026, 2017.
- [47] A. Eltner, D. Mader, N. Szopos, B. Nagy, J. Grundmann, and L. Bertalan, "Using thermal and RGB UAV imagery to measure surface flow velocities," *The International Archives* of the Photogrammetry, Remote Sensing and Spatial Information Sciences, vol. 43, pp. 717–722, 2021.
- [48] S. Wang, M. Garcia, P. Bauer-Gottwein et al., "High spatial resolution monitoring land surface energy, water and CO2 fluxes from an Unmanned Aerial System," *Remote Sensing of Environment*, vol. 229, pp. 14–31, 2019.
- [49] T. De Kerf, J. Gladines, S. Sels, and S. Vanlanduit, "Oil spill detection using machine learning and infrared images," *Remote Sensing*, vol. 12, no. 24, p. 4090, 2020.
- [50] K. H. Cheng, J. J. Jiao, X. Luo, and S. Yu, "Effective coastal Escherichia coli monitoring by unmanned aerial vehicles (UAV) thermal infrared images," *Water Research*, vol. 222, Article ID 118900, 2022.

- [51] L. Goddijn-Murphy, B. J. Williamson, J. McIlvenny, P. Corradi, and P. Corradi, "Using a UAV thermal infrared camera for monitoring floating marine plastic litter," *Remote Sensing*, vol. 14, no. 13, p. 3179, 2022.
- [52] J. Bunker, R. M. Nagisetty, and J. Crowley, "sUAS remote sensing to evaluate geothermal seep interactions with the yellowstone river, Montana, USA," *Remote Sensing*, vol. 13, no. 2, pp. 163–228, 2021.
- [53] V. Döpper, T. Gränzig, B. Kleinschmit, and M. Förster, "Challenges in UAS-based TIR imagery processing: image alignment and uncertainty quantification," *Remote Sensing*, vol. 12, no. 10, p. 1552, 2020.
- [54] S. Labbé, V. Lebourgeois, A. Jolivot, and R. Marti, "Thermal infra-red remote sensing for water stress estimation in agriculture - institut National de Recherche en Agriculture," *Options Méditerranéennes Série B. Etudes et Recherches*, vol. 67, 2012.
- [55] G. Ludeno, I. Catapano, A. Renga, A. R. Vetrella, G. Fasano, and F. Soldovieri, "Assessment of a micro-UAV system for microwave tomography radar imaging," *Remote Sensing of Environment*, vol. 212, pp. 90–102, Jun. 2018.
- [56] G. Fasano, A. Renga, A. R. Vetrella, G. Ludeno, I. Catapano, and F. Soldovieri, "Proof of concept of micro-UAV-based radar imaging," in *Proceedings of the International Conference on Unmanned Aircraft Systems*, pp. 1316–1323, Miami, FL, USA, July 2017.
- [57] F. Bandini, J. Jakobsen, D. Olesen, J. A. Reyna-Gutierrez, and P. Bauer-Gottwein, "Measuring water level in rivers and lakes from lightweight Unmanned Aerial Vehicles," *Journal* of Hydrology, vol. 548, pp. 237–250, 2017.
- [58] P. A. Rosen, S. Hensley, K. Wheeler et al., "UAVSAR: a new NASA airborne SAR system for science and technology research," in *Proceedings of the 2006 IEEE Conference on Radar*, pp. 22–29, Verona, NY, USA, April 2006.
- [59] D. P. S. Bekaert, C. E. Jones, K. An, and M. H. Huang, "Exploiting UAVSAR for a comprehensive analysis of subsidence in the Sacramento Delta," *Remote Sensing of Environment*, vol. 220, pp. 124–134, 2019.
- [60] M. M. Miller, C. E. Jones, S. S. Sangha, and D. P. Bekaert, "Rapid drought-induced land subsidence and its impact on the California aqueduct," *Remote Sensing of Environment*, vol. 251, Article ID 112063, 2020.
- [61] J. Jones, C. E. Jones, and D. P. S. Bekaert, "Value of InSAR for monitoring land subsidence to support water management in the san joaquin valley, California," *JAWRA Journal of the American Water Resources Association*, 2021.
- [62] M. A. Remy, K. A. C. De MacEdo, and J. R. Moreira, "The first UAV-based P- and X-band interferometric SAR system," in *Proceedings of the 2012 IEEE International Geoscience and Remote Sensing Symposium*, pp. 5041–5044, Munich, Germany, July 2012.
- [63] K. Ouchi, "Recent trend and advance of synthetic aperture radar with selected topics," *Remote Sensing*, vol. 5, no. 2, pp. 716–807, 2013.
- [64] K. Wu, G. A. Rodriguez, M. Zajc et al., "A new drone-borne GPR for soil moisture mapping," *Remote Sensing of Envi*ronment, vol. 235, p. 111456, 2019.
- [65] R. O. R. Jenssen and S. K. Jacobsen, "Measurement of snow water equivalent using drone-mounted ultra-wide-band radar," *Remote Sensing*, vol. 13, no. 13, p. 2610, 2021.
- [66] E. M. McIntyre and A. J. Gasiewski, "An ultra-lightweight L-band digital Lobe-Differencing Correlation Radiometer (LDCR) for airborne UAV SSS mapping," 2007 IEEE

International Geoscience and Remote Sensing Symposium, pp. 1095–1097, 2007.

- [67] R. Deurloo, L. Bastos, and M. Bos, "On the use of UAVs for strapdown airborne gravimetry," *Geodesy for Planet Earth*, vol. 136, pp. 255–261, 2012.
- [68] R. P. Middlemiss, A. Samarelli, D. J. Paul, J. Hough, S. Rowan, and G. D. Hammond, "Measurement of the Earth tides with a MEMS gravimeter," *Nature*, vol. 531, no. 7596, pp. 614–617, 2016.
- [69] C. A. Lin, K. W. Chiang, and C. Y. Kuo, "Development of INS/GNSS UAV-borne vector gravimetry system," *IEEE Geoscience and Remote Sensing Letters*, vol. 14, no. 5, pp. 759–763, 2017.
- [70] B. Li, J. Hou, D. Li et al., "Application of LiDAR UAV for high-resolution flood modelling," *Water Resources Man*agement, vol. 35, no. 5, pp. 1433–1447, 2021.
- [71] S. R. Rogers, I. Manning, and W. Livingstone, "Comparing the spatial accuracy of digital surface models from four unoccupied aerial systems: photogrammetry versus LiDAR," *Remote Sensing*, vol. 12, no. 17, p. 2806, 2020.
- [72] T. J. Pingel, A. Saavedra, and L. Cobo, "Deriving land and water surface elevations in the northeastern yucatán peninsula using PPK GPS and UAV-based structure from motion," *Papers in Applied Geography*, vol. 7, no. 3, pp. 294–315, 2021.
- [73] G. Mandlburger, M. Pfennigbauer, R. Schwarz, S. Flöry, and L. Nussbaumer, "Concept and performance evaluation of a novel UAV-borne topo-bathymetric LiDAR sensor," *Remote Sensing*, vol. 12, no. 6, p. 986, 2020.
- [74] V. Raimondi, L. Palombi, D. Lognoli, A. Masini, and E. Simeone, "Experimental tests and radiometric calculations for the feasibility of fluorescence LIDAR-based discrimination of oil spills from UAV," *International Journal of Applied Earth Observation and Geoinformation*, vol. 61, pp. 46–54, 2017.
- [75] Z. Duan, Y. Li, J. Wang, G. Zhao, and S. Svanberg, "Aquatic environment monitoring using a drone-based fluorosensor," *Applied Physics B*, vol. 125, no. 6, p. 108, 2019.
- [76] D. Baek, I. W. Seo, J. S. Kim, and J. M. Nelson, "UAV-based measurements of spatio-temporal concentration distributions of fluorescent tracers in open channel flows," *Advances in Water Resources*, vol. 127, pp. 76–88, 2019.
- [77] H. Eisenbeiss, *UAV photogrammetry*, ETH Zürich, Switzerland, 2009.
- [78] F. Nex and F. Remondino, "UAV for 3D mapping applications: a review," *Applied Geomatics*, vol. 6, no. 1, pp. 1–15, 2014.
- [79] K. Jain, "How photogrammetric software works: a perspective based on UAV's exterior orientation parameters," *J. Indian Soc. Remote Sens.*vol. 49, no. 3, pp. 641–649, 2021.
- [80] S. I. Deliry and U. Avdan, "Accuracy of unmanned aerial systems photogrammetry and structure from motion in surveying and mapping: a review," *J. Indian Soc. Remote Sens*, vol. 49, no. 8, pp. 1997–2017, 2021.
- [81] A. Al-Kaff, D. Martín, F. García, A. de la Escalera, and J. Armingol, "Survey of computer vision algorithms and applications for unmanned aerial vehicles," *Expert Systems with Applications*, vol. 92, 2018.
- [82] S. Zazo, J. L. Molina, and P. Rodríguez-Gonzálvez, "Analysis of flood modeling through innovative geomatic methods," *Journal of Hydrology*, vol. 524, pp. 522–537, 2015.
- [83] W. A. Marcus and M. A. Fonstad, "Optical remote mapping of rivers at sub-meter resolutions and watershed extents,"

Earth Surface Processes and Landforms, vol. 33, no. 1, pp. 4–24, 2008.

- [84] R. D'Andrimont, C. Marlier, and P. D. Sensing, "Hyperspatial and multi-source water body mapping: a framework to handle heterogeneities from observations and targets over large areas," *Remote Sensing*, 2017.
- [85] J. Jensen, "Remote sensing of the environment: an earth resource perspective 2/e," 2009.
- [86] P. J. Hardin and R. R. Jensen, "Small-scale unmanned aerial vehicles in environmental remote sensing: challenges and opportunities," *GIScience and Remote Sensing*, vol. 48, no. 1, pp. 99–111, 2013.
- [87] C. Gomez and H. Purdie, "UAV- based photogrammetry and geocomputing for hazards and disaster risk monitoring – a review," *Geoenvironmental Disasters*, vol. 3, no. 1, pp. 23–11, 2016.
- [88] S. Śledź, M. W. Ewertowski, and J. Piekarczyk, "Applications of unmanned aerial vehicle (UAV) surveys and Structure from Motion photogrammetry in glacial and periglacial geomorphology," *Geomorphology*, vol. 378, Article ID 107620, 2021.
- [89] A. Bhardwaj, L. Sam, A. Akankshad, F. J. Martín-Torres, and R. Kumar, "UAVs as remote sensing platform in glaciology: present applications and future prospects," *Remote Sensing* of Environment, vol. 175, pp. 196–204, 2016.
- [90] W. McDonald, "Drones in urban stormwater management: a review and future perspectives," Urban Water Journal, vol. 16, no. 7, pp. 505–518, 2019.
- [91] A. Carrio, C. Sampedro, A. Rodriguez-Ramos, and P. Campoy, "A review of deep learning methods and applications for unmanned aerial vehicles," *Journal of Sensors*, vol. 2017, pp. 1–13, Article ID 3296874, 2017.
- [92] D. Wu, R. Li, F. Zhang, and J. Liu, "A review on drone-based harmful algae blooms monitoring," *Environmental Monitoring and Assessment*, vol. 191, no. 4, p. 211, 2019.
- [93] S. Jiang, C. Jiang, and W. Jiang, "Efficient structure from motion for large-scale UAV images: a review and a comparison of SfM tools," *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 167, pp. 230–251, 2020.
- [94] D. S. Rhee, Y. D. Kim, B. Kang, and D. Kim, "Applications of unmanned aerial vehicles in fluvial remote sensing: an overview of recent achievements," *KSCE Journal of Civil Engineering*, vol. 22, no. 2, pp. 588–602, 2017.
- [95] P. K. Mishra and A. Rai, "Role of unmanned aerial systems for natural resource management," J. Indian Soc. Remote Sens.vol. 49, no. 3, pp. 671–679, 2021.
- [96] M. Chung, C. Detweiler, M. Hamilton, J. Higgins, J. P. Ore, and S. Thompson, "Obtaining the thermal structure of lakes from the air," *Water*, vol. 7, no. 11, pp. 6467–6482, 2015.
- [97] T.-C. Su, H.-T. Chou, G. Pajares Martinsanz, R. Müller, A. Lucieer, and P. S. Thenkabail, "Application of multispectral sensors carried on unmanned aerial vehicle (UAV) to trophic state mapping of small reservoirs: a case study of tain-Pu reservoir in kinmen, taiwan," *Remote Sensing*, vol. 7, no. 8, pp. 10078–10097, 2015.
- [98] E. Ridolfi and P. Manciola, "Water level measurements from drones: a pilot case study at a dam site," *Water*, vol. 10, no. 3, p. 297, 2018.
- [99] H. Doi, Y. Akamatsu, Y. Watanabe et al., "Water sampling for environmental DNA surveys by using an unmanned aerial vehicle," *Limnology and Oceanography: Methods*, vol. 15, no. 11, pp. 939–944, 2017.
- [100] K. Song, A. Brewer, S. Ahmadian, A. Shankar, C. Detweiler, and A. J. Burgin, "Using unmanned aerial vehicles to sample

aquatic ecosystems," *Limnology and Oceanography: Methods*, vol. 15, no. 12, pp. 1021–1030, 2017.

- [101] D. Chabot, C. Dillon, A. Shemrock, N. Weissflog, and E. Sager, "An object-based image analysis workflow for monitoring shallow-water aquatic vegetation in multispectral drone imagery," *ISPRS International Journal of Geo-Information*, vol. 7, no. 8, p. 294, 2018.
- [102] A. Terada, "Water sampling using a drone at Yugama crater lake, Kusatsu-Shirane volcano, Japan," *Earth Planets and Space*, vol. 70, no. 1, pp. 1–9, 2018.
- [103] J. Wang, T. Shi, D. Yu et al., "Ensemble machine-learningbased framework for estimating total nitrogen concentration in water using drone-borne hyperspectral imagery of emergent plants: a case study in an arid oasis, NW China," *Environmental Pollution*, vol. 266, p. 115412, 2020.
- [104] B. Fu, P. Xu, Y. Wang, K. Yan, and S. Chaudhary, "Assessment of the ecosystem services provided by ponds in hilly areas," *Science of the Total Environment*, vol. 642, pp. 979–987, 2018.
- [105] A. M. Tomczyk and M. W. Ewertowski, "UAV-based remote sensing of immediate changes in geomorphology following a glacial lake outburst flood at the Zackenberg river, northeast Greenland," *Journal of Maps*, vol. 16, no. 1, pp. 86–100, 2020.
- [106] A. S. Woodget, P. E. Carbonneau, F. Visser, and I. P. Maddock, "Quantifying submerged fluvial topography using hyperspatial resolution UAS imagery and structure from motion photogrammetry," *Earth Surface Processes and Landforms*, vol. 40, no. 1, pp. 47–64, 2015.
- [107] S. W. Jang, H. J. Yoon, S. N. Kwak, B. Y. Sohn, S. G. Kim, and D. H. Kim, "Algal bloom monitoring using UAVs imagery," vol. 34350, pp. 30–33, 2016.
- [108] H.-M. Kim, H. J. Yoon, S. W. Jang et al., "Application of unmanned aerial vehicle imagery for algal bloom monitoring in river basin," *International Journal of Control and Automation*, vol. 9, no. 12, pp. 203–220, 2016.
- [109] S. Coveney and K. Roberts, "Lightweight UAV digital elevation models and orthoimagery for environmental applications: data accuracy evaluation and potential for river flood risk modelling," *International Journal of Remote Sensing*, vol. 38, no. 8–10, pp. 3159–3180, 2017.
- [110] J. Langhammer, J. Bernsteinová, and J. Miřijovský, "Building a high-precision 2D hydrodynamic flood model using UAV photogrammetry and sensor network monitoring," *Water*, vol. 9, no. 11, p. 861, 2017.
- [111] A. S. Woodget and R. Austrums, "Subaerial gravel size measurement using topographic data derived from a UAV-SfM approach," *Earth Surface Processes and Landforms*, vol. 42, no. 9, pp. 1434–1443, 2017.
- [112] J. Langhammer and T. Vacková, "Detection and mapping of the geomorphic effects of flooding using UAV photogrammetry," *Pure and Applied Geophysics*, vol. 175, no. 9, pp. 3223–3245, 2018.
- [113] S. Hemmelder, W. Marra, H. Markies, and S. M. De Jong, "Monitoring river morphology & bank erosion using UAV imagery – a case study of the river Buëch, Hautes-Alpes, France," *International Journal of Applied Earth Observation* and Geoinformation, vol. 73, pp. 428–437, 2018.
- [114] F. Bandini, T. P. Sunding, J. Linde et al., "Unmanned Aerial System (UAS) observations of water surface elevation in a small stream: comparison of radar altimetry, LIDAR and photogrammetry techniques," *Remote Sensing of Environment*, vol. 237, p. 111487, 2020.
- [115] M. Cassel, H. Piegay, G. Fantino et al., "Comparison of ground-based and UAV a-UHF artificial tracer mobility

monitoring methods on a braided river," *Earth Surface Processes and Landforms*, vol. 45, no. 5, pp. 1123–1140, 2020.

- [116] J. Lejot, C. Delacourt, H. Piégay, T. Fournier, M. L. Trémélo, and P. Allemand, "Very high spatial resolution imagery for channel bathymetry and topography from an unmanned mapping controlled platform," *Earth Surface Processes and Landforms*, vol. 32, no. 11, pp. 1705–1725, 2007.
- [117] D. Beene, S. Zhang, and G. Paulus, "Workflow for hydrologic modelling with sUAS-acquired aerial imagery," *Geocarto International*, vol. 36, no. 12, pp. 1346–1364, 2021.
- [118] E. J. Kim, S. H. Nam, J. W. Koo, and T. M. Hwang, "Hybrid approach of unmanned aerial vehicle and unmanned surface vehicle for assessment of chlorophyll-a imagery using spectral indices in stream, South Korea," *Water*, vol. 13, no. 14, p. 1930, 2021.
- [119] S. Rathinam, Z. W. Kim, and R. Sengupta, "Vision-based monitoring of locally linear structures using an unmanned aerial vehicle," *Journal of Infrastructure Systems*, vol. 14, no. 1, pp. 52–63, 2008.
- [120] R. Dunford, K. Michel, M. Gagnage, H. Piégay, and M. L. Trémelo, "Potential and constraints of Unmanned Aerial Vehicle technology for the characterization of Mediterranean riparian forest," *International Journal of Remote Sensing*, vol. 30, no. 19, pp. 4915–4935, 2009.
- [121] D. Vericat, J. Brasington, J. Wheaton, and M. Cowie, "Accuracy assessment of aerial photographs acquired using lighter-than-air blimps: LOW-cost tools for mapping river corridors," *River Research and Applications*, vol. 25, no. 8, pp. 985–1000, 2009.
- [122] K. F. Flynn and S. C. Chapra, "Remote sensing of submerged aquatic vegetation in a shallow non-turbid river using an unmanned aerial vehicle," *Remote Sensing*, vol. 6, no. 12, pp. 12815–12836, 2014.
- [123] M. Casado, R. Gonzalez, T. Kriechbaumer, and A Veal, "Automated identification of river hydromorphological features using UAV high resolution aerial imagery," *Sensors*, vol. 15, no. 11, pp. 27969–27989, 2015.
- [124] J. Miřijovský and J. Langhammer, "Multitemporal monitoring of the morphodynamics of a mid-mountain stream using UAS photogrammetry," *Remote Sensing*, vol. 7, no. 7, pp. 8586–8609, 2015.
- [125] A. D. Tamminga, B. C. Eaton, and C. H. Hugenholtz, "UASbased remote sensing of fluvial change following an extreme flood event," *Earth Surface Processes and Landforms*, vol. 40, no. 11, pp. 1464–1476, 2015.
- [126] S. Pearce, R. Ljubicic, S. Pena-Haro et al., "An evaluation of image velocimetry techniques under low flow conditions and high seeding densities using unmanned aerial systems," *Remote Sensing*, vol. 12, no. 2, p. 232, 2020.
- [127] F. Tauro, A. Petroselli, and E. Arcangeletti, "Assessment of drone-based surface flow observations," *Hydrological Processes*, vol. 30, no. 7, pp. 1114–1130, 2016.
- [128] D. Chabot and D. M. Bird, "Small unmanned aircraft: precise and convenient new tools for surveying wetlands," *Journal of Unmanned Vehicle Systems*, vol. 1, no. 1, pp. 15–24, 2013.
- [129] J. V. Marcaccio, C. E. Markle, and P. Chow-Fraser, "Use of fixed-wing and multi-rotor unmanned aerial vehicles to map dynamic changes in a freshwater marsh," *Journal of Unmanned Vehicle Systems*, vol. 4, no. 3, pp. 193–202, 2016.
- [130] M. A. Boon, R. Greenfield, and S. Tesfamichael, "Wetland assessment using unmanned aerial vehicle (UAV) photogrammetry," *ISPRS - International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. 41, pp. 781–788, 2016.

- [131] X. Zhang, C. E. Jones, T. Oliver-Cabrera, M. Simard, and S. Fagherazzi, "Using rapid repeat SAR interferometry to improve hydrodynamic models of flood propagation in coastal wetlands," *Advances in Water Resources*, vol. 159, Article ID 104088, 2022.
- [132] F. Tauro, M. Porfiri, and S. Grimaldi, "Surface flow measurements from drones," *Journal of Hydrology*, vol. 540, pp. 240–245, 2016.
- [133] D. Strelnikova, G. Paulus, S. Kafer et al., "Drone-based optical measurements of heterogeneous surface velocity fields around fish passages at hydropower dams," *Remote Sensing*, vol. 12, no. 3, p. 384, 2020.
- [134] M. Kalacska, G. L. Chmura, O. Lucanus, D. Bérubé, and J. P. Arroyo-Mora, "Structure from motion will revolutionize analyses of tidal wetland landscapes," *Remote Sensing of Environment*, vol. 199, pp. 14–24, 2017.
- [135] Y. Zhang, L. Wu, L. Deng, and B. Ouyang, "Retrieval of water quality parameters from hyperspectral images using a hybrid feedback deep factorization machine model," *Water Research*, vol. 204, p. 117618, 2021.
- [136] H. Xia, W. Zhao, A. Li, J. Bian, and Z. Zhang, "Subpixel inundation mapping using landsat-8 OLI and UAV data for a wetland region on the zoige plateau, China," *Remote Sensing*, vol. 9, no. 1, p. 31, 2017.
- [137] M. Detert, E. D. Johnson, and V. Weitbrecht, "Proof-ofconcept for low-cost and non-contact synoptic airborne river flow measurements," *International Journal of Remote Sensing*, vol. 38, no. 8–10, pp. 2780–2807, 2017.
- [138] G. Kaplan and U. Avdan, "Monthly analysis of wetlands dynamics using remote sensing data," *ISPRS International Journal of Geo-Information*, vol. 7, no. 10, p. 411, 2018.
- [139] W. W. Immerzeel, P. Kraaijenbrink, J. Shea et al., "Highresolution monitoring of Himalayan glacier dynamics using unmanned aerial vehicles," *Remote Sensing of Environment*, vol. 150, pp. 93–103, 2014.
- [140] R. Ramsankaran, P. J. Navinkumar, A. Dashora, and A. V. Kulkarni, "UAV-based survey of glaciers in himalayas: challenges and recommendations," *J. Indian Soc. Remote Sens.*vol. 49, no. 5, pp. 1171–1187, 2021.
- [141] N. Karimi, S. Sheshangosht, and R. Roozbahani, "Highresolution monitoring of debris-covered glacier mass budget and flow velocity using repeated UAV photogrammetry in Iran," *Geomorphology*, vol. 389, Article ID 107855, 2021.
- [142] D. C. Leal-Alves, J. Weschenfelder, J. C. Dominguez Almeida, M. da Guia Albuquerque, J. M. de Almeida Espinoza, and B. A. Gonzaga, "Unmanned aerial vehicle and structure from motion approach for flood assessment in coastal channels," *Journal of Coastal Research*, vol. 95, no. sp1, pp. 1162–1166, 2020.
- [143] R. McEliece, S. Hinz, J. M. Guarini, and J. Coston-Guarini, "Evaluation of nearshore and offshore water quality assessment using UAV multispectral imagery," *Remote Sensing*, vol. 12, no. 14, p. 2258, 2020.
- [144] A. Papakonstantinou, M. Batsaris, S. Spondylidis, and K. Topouzelis, "A citizen science unmanned aerial system data acquisition protocol and deep learning techniques for the automatic detection and mapping of marine litter concentrations in the coastal zone," *Drones*, vol. 5, no. 1, p. 6, 2021.
- [145] D. C. Leal-Alves, J. Weschenfelder, M. d. G. Albuquerque, J. M. d. A. Espinoza, M. Ferreira-Cravo, and L. P. M. d. Almeida, "Digital elevation model generation using UAV-SfM photogrammetry techniques to map sea-

level rise scenarios at Cassino Beach, Brazil," SN Applied Sciences, vol. 2, no. 12, pp. 2181–2219, 2020.

- [146] N. Long, B. Millescamps, B. Guillot, F. Pouget, and X. Bertin, "Monitoring the topography of a dynamic tidal inlet using UAV imagery," *Remote Sensing*, vol. 8, no. 5, p. 387, 2016.
- [147] C. Cillero Castro, J. A. Dominguez Gomez, J. Delgado Martin et al., "An UAV and satellite multispectral data approach to monitor water quality in small reservoirs," *Remote Sensing*, vol. 12, no. 9, p. 1514, 2020.
- [148] S. L. Murfitt, B. M. Allan, A. Bellgrove, A. Rattray, M. A. Young, and D. Ierodiaconou, "Applications of unmanned aerial vehicles in intertidal reef monitoring," *Scientific Reports*, vol. 7, no. 1, pp. 10259–10311, 2017.
- [149] K. Appeaning Addo, P. N. Jayson-Quashigah, S. N. A. Codjoe, and F. Martey, "Drone as a tool for coastal flood monitoring in the Volta Delta, Ghana," *Geoenvironmental Disasters*, vol. 5, no. 1, p. 17, 2018.
- [150] F. Xu, Z. Gao, X. Jiang et al., "A UAV and S2A data-based estimation of the initial biomass of green algae in the South Yellow Sea," *Marine Pollution Bulletin*, vol. 128, pp. 408–414, 2018.
- [151] P. N. Jayson-Quashigah, K. Appeaning Addo, B. Amisigo, and G. Wiafe, "Assessment of short-term beach sediment change in the Volta Delta coast in Ghana using data from Unmanned Aerial Vehicles (Drone)," Ocean & Coastal Management, vol. 182, Article ID 104952, 2019.
- [152] Z. Jiao, G. Jia, and Y. Cai, "A new approach to oil spill detection that combines deep learning with unmanned aerial vehicles," *Computers & Industrial Engineering*, vol. 135, pp. 1300–1311, 2019.
- [153] K. Johansen, A. F. Dunne, Y. H. Tu, B. H. Jones, and M. F. McCabe, "Monitoring coastal water flow dynamics using sub-daily high-resolution SkySat satellite and UAVbased imagery," *Water Research*, vol. 219, Article ID 118531, 2022.
- [154] K. Lamsters, J. Karušs, M. Krievāns, and J. Ješkins, "Highresolution orthophoto map and digital surface models of the largest Argentine Islands (the Antarctic) from unmanned aerial vehicle photogrammetry," *Journal of Maps*, vol. 16, no. 2, pp. 335–347, 2020.
- [155] M. Shahbazi, J. Théau, and P. Ménard, "Recent applications of unmanned aerial imagery in natural resource management," *GIScience and Remote Sensing*, vol. 51, no. 4, pp. 339–365, 2014.
- [156] J. Bunker, R. M. Nagisetty, and J. Crowley, "sUAS remote sensing to evaluate geothermal seep interactions with the yellowstone river, Montana, USA," *Remote Sensing*, vol. 13, no. 2, p. 163, 2021.
- [157] M. C. Harvey, J. V. Rowland, and K. M. Luketina, "Drone with thermal infrared camera provides high resolution georeferenced imagery of the Waikite geothermal area, New Zealand," *Journal of Volcanology and Geothermal Research*, vol. 325, pp. 61–69, 2016.
- [158] J. Kuhn, R. Casas-Mulet, J. Pander, and J. Geist, "Assessing stream thermal heterogeneity and cold-water patches from UAV-based imagery: a matter of classification methods and metrics," *Remote Sensing*, vol. 13, no. 7, p. 1379, 2021.
- [159] M. Lega and R. M. A. Napoli, "Aerial infrared thermography in the surface waters contamination monitoring," *Desalination and Water Treatment*, vol. 23, no. 1–3, pp. 141–151, 2010.
- [160] M. Lega, J. Kosmatka, C. Ferrara, F. Russo, R. M. A. Napoli, and G. Persechino, "Using advanced aerial platforms and infrared thermography to track environmental

contamination," *Environmental Forensics*, vol. 13, no. 4, pp. 332–338, 2012.

- [161] W. Wilkowski, M. Lisowski, M. Wyszyński, and D. Wierzbicki, "The use of unmanned aerial vehicles (drones) to determine the shoreline of natural watercourses," *Journal of Water and Land Development*, vol. 35, no. 1, pp. 259–264, Dec. 2017.
- [162] T. Templin, D. Popielarczyk, and R. Kosecki, "Application of low-cost fixed-wing UAV for inland lakes shoreline investigation," *Pageoph Topical Volumes*, pp. 123–143, 2018.
- [163] A. Medvedev, N. Telnova, N. Alekseenko et al., "UAV-derived data application for environmental monitoring of the coastal area of lake sevan, Armenia with a changing water level," *Remote Sensing*, vol. 12, no. 22, p. 3821, 2020.
- [164] R. C. Templeton, E. R. Vivoni, L. A. Mendez-Barroso et al., "High-resolution characterization of a semiarid watershed: implications on evapotranspiration estimates," *Journal of Hydrology*, vol. 509, pp. 306–319, 2014.
- [165] C. Spence and S. Mengistu, "Deployment of an unmanned aerial system to assist in mapping an intermittent stream," *Hydrological Processes*, vol. 30, no. 3, pp. 493–500, 2016.
- [166] N. Pineux, J. Lisein, G. Swerts et al., "Can DEM time series produced by UAV be used to quantify diffuse erosion in an agricultural watershed?" *Geomorphology*, vol. 280, pp. 122– 136, 2017.
- [167] F. Argüello, D. B. Heras, A. S. Garea, and P. Quesada-Barriuso, "Watershed monitoring in galicia from UAV multispectral imagery using advanced texture methods," *Remote Sensing*, vol. 13, no. 14, p. 2687, 2021.
- [168] J. L. Carrivick and M. W. Smith, "Fluvial and aquatic applications of Structure from Motion photogrammetry and unmanned aerial vehicle/drone technology," *WIREs Water*, vol. 6, no. 1, 2019.
- [169] Y. Anker, Y. Hershkovitz, E. Ben Dor, and A. Gasith, "Application of aerial digital photography for macrophyte cover and composition survey in small rural streams," *River Research and Applications*, vol. 30, no. 7, pp. 925–937, 2014.
- [170] S. Nuske, S. Choudhury, S. Jain et al., "Autonomous exploration and motion planning for an unmanned aerial vehicle navigating rivers," *Journal of Field Robotics*, vol. 32, no. 8, pp. 1141–1162, 2015.
- [171] M. R. Casado, R. B. Gonzalez, T. Kriechbaumer, and A. Veal, "Automated identification of river hydromorphological features using UAV high resolution aerial imagery," *Sensors*, vol. 15, no. 11, pp. 27969–27989, 2015.
- [172] M. Calle, P. Alho, and G. Benito, "Monitoring ephemeral river changes during floods with SfM photogrammetry," *Journal of Iberian Geology*, vol. 44, no. 3, pp. 355–373, 2018.
- [173] S. Kubota, Y. Kawai, and R. Kadotani, "Accuracy validation of point clouds of UAV photogrammetry and its application for river management," *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. 42, pp. 195–199, 2017.
- [174] C. S. Zhao, C. Zhang, S. Yang et al., "Calculating e-flow using UAV and ground monitoring," *Journal of Hydrology*, vol. 552, pp. 351–365, 2017.
- [175] S. Zazo, P. Rodríguez-Gonzálvez, J. L. Molina, D. González-Aguilera, C. A. Agudelo-Ruiz, and D. Hernández-López, "Flood hazard assessment supported by reduced cost aerial precision photogrammetry," *Remote Sensing*, vol. 10, no. 10, p. 1566, 2018.
- [176] C. Flener, M. Vaaja, A. Jaakkola et al., "Seamless mapping of river channels at high resolution using mobile LiDAR and

UAV-photography," *Remote Sensing*, vol. 5, no. 12, pp. 6382-6407, 2013.

- [177] S. Scherer, J. Rehder, S. Achar et al., "River mapping from a flying robot: state estimation, river detection, and obstacle mapping," *Autonomous Robots*, vol. 33, no. 1–2, pp. 189–214, 2012.
- [178] E. Ferreira, J. Chandler, R. Wackrow, and K. Shiono, "Automated extraction of free surface topography using SfM-MVS photogrammetry," *Flow Measurement and Instrumentation*, vol. 54, pp. 243–249, 2017.
- [179] A. Eltner, P. O. Bressan, T. Akiyama, W. N. Goncalves, and J. Marcato Junior, "Using deep learning for automatic water stage measurements," *Water Resources Research*, vol. 57, no. 3, 2021.
- [180] F. Bahmanpouri, A. Eltner, S. Barbetta, L. Bertalan, and T. Moramarco, "Estimating the average river cross section velocity by observing only one surface velocity value and calibrating the entropic parameter," *Water Resources Research*, vol. 58, 2022.
- [181] P. Koutalakis, O. Tzoraki, and G. Zaimes, "UAVs for hydrologic scopes: application of a low-cost UAV to estimate surface water velocity by using three different image-based methods," *Drones*, vol. 3, no. 1, p. 14, 2019.
- [182] E. Tabacchi, L. Lambs, H. Guilloy, A.-M. Planty-Tabacchi, and E. Muller, "Impacts of riparian vegetation on hydrological processes," *Hydrological Processes*, vol. 14, no. 16–17, pp. 2959–2976, 2000.
- [183] A. Michez, H. Piégay, J. Lisein, H. Claessens, and P. Lejeune, "Classification of riparian forest species and health condition using multi-temporal and hyperspatial imagery from unmanned aerial system," *Environmental Monitoring and Assessment*, vol. 188, no. 3, pp. 146–219, 2016.
- [184] J. H. Yi, C. J. Shin, K. H. Ryu, and W. M. Jeong, "Bathymetry survey using an aerial image for a very shallow water region," in *Proceedings of the Ocean. 2016 MTS/IEEE Monterey*, Monterey, CA, USA, September 2016.
- [185] P. E. Carbonneau, S. N. Lane, and N. Bergeron, "Feature based image processing methods applied to bathymetric measurements from airborne remote sensing in fluvial environments," *Earth Surface Processes and Landforms*, vol. 31, no. 11, pp. 1413–1423, 2006.
- [186] L. Javernick, J. Brasington, and B. Caruso, "Modeling the topography of shallow braided rivers using Structure-from-Motion photogrammetry," *Geomorphology*, vol. 213, pp. 166–182, 2014.
- [187] Z. Pan, C. Glennie, P. Hartzell, J. C. Fernandez-Diaz, C. Legleiter, and B. Overstreet, "Performance assessment of high resolution airborne full waveform LiDAR for shallow river bathymetry," *Remote Sensing*, vol. 7, no. 5, pp. 5133–5159, 2015.
- [188] C. Shintani and M. A. Fonstad, "Comparing remote-sensing techniques collecting bathymetric data from a gravel-bed river," *International Journal of Remote Sensing*, vol. 38, no. 8–10, pp. 2883–2902, 2017.
- [189] J. He, J. Lin, M. Ma, and X. Liao, "Mapping topo-bathymetry of transparent tufa lakes using UAV-based photogrammetry and RGB imagery," *Geomorphology*, vol. 389, p. 107832, 2021.
- [190] E. Maltby and T. M. T. Barker, *The Wetlands Handbook*, Wiley Online Library, Hoboken, NJ. USA, 2009.
- [191] D. Chabot, C. Dillon, A. Shemrock, N. Weissflog, and E. P. S. Sager, "An object-based image analysis workflow for monitoring shallow-water aquatic vegetation in

multispectral drone imagery," *ISPRS International Journal of Geo-Information*, vol. 7, no. 8, p. 294, 2018.

- [192] S. Dingman, *Physical Hydrology*, Waveland press, Grove, IL, USA, 2015.
- [193] P. Dobriyal, A. Qureshi, R. Badola, and S. A. Hussain, "A review of the methods available for estimating soil moisture and its implications for water resource management," *Journal of Hydrology*, vol. 458, pp. 110–117, 2012.
- [194] L. Hassan-Esfahani, A. Torres-Rua, A. Jensen, and M. McKee, "Assessment of surface soil moisture using highresolution multi-spectral imagery and artificial neural networks," *Remote Sensing*, vol. 7, no. 3, pp. 2627–2646, 2015.
- [195] X. Ge, J. Wang, J. Ding et al., "Combining UAV-based hyperspectral imagery and machine learning algorithms for soil moisture content monitoring," *PeerJ*, vol. 7, 2019.
- [196] R. Acevo-Herrera, A. Aguasca, X. Bosch-Lluis et al., "Design and first results of an UAV-borne L-band radiometer for multiple monitoring purposes," *Remote Sensing*, vol. 2, no. 7, pp. 1662–1679, 2010.
- [197] T. C. Su and T. Ching, "A study of a matching pixel by pixel (MPP) algorithm to establish an empirical model of water quality mapping, as based on unmanned aerial vehicle (UAV) images," *International Journal of Applied Earth Observation and Geoinformation*, vol. 58, pp. 213–224, 2017.
- [198] C. Zeng, M. Richardson, and D. J. King, "The impacts of environmental variables on water reflectance measured using a lightweight unmanned aerial vehicle (UAV)-based spectrometer system," *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 130, pp. 217–230, 2017.
- [199] Y. Hanting, "Evaluation of water quality based on UAV images and the IMP-MPP algorithm | Article Information | J-GLOBAL," *Ecological Informatics*, vol. 61, 2021.
- [200] C. Detweiler, J. P. Ore, D. Anthony, S. Elbaum, A. Burgin, and A. Lorenz, "Environmental reviews and case studies: bringing unmanned aerial systems closer to the environment," *Environmental Practice*, vol. 17, no. 3, pp. 188–200, 2015.
- [201] J. P. Ore, S. Elbaum, A. Burgin, and C. Detweiler, "Autonomous aerial water sampling," *Journal of Field Robotics*, vol. 32, no. 8, pp. 1095–1113, 2015.
- [202] C. Koparan and A. B. Koc, "Unmanned Aerial Vehicle (UAV) assisted water sampling," in *Proceedings of the 2016 Am. Soc. Agric. Biol. Eng. Annu. Int. Meet. ASABE*, p. 1, New York, NY, USA, September 2016.
- [203] P. Rodrigues, "An open-source watertight unmanned aerial vehicle for water quality monitoring," in *Proceedings of the* OCEANS 2015 - MTS/IEEE Washington, Washington, DC, USA, Februar 2016.
- [204] X. Yang, T. Wang, J. Liang, G. Yao, and M. Liu, "Survey on the novel hybrid aquatic-aerial amphibious aircraft: aquatic unmanned aerial vehicle (AquaUAV)," *Progress in Aerospace Sciences*, vol. 74, pp. 131–151, 2015.
- [205] J. Ore and C. Detweiler, "Sensing water properties at precise depths from the air," *Journal of Field Robotics*, vol. 35, no. 8, pp. 1205–1221, 2018.
- [206] J. Benson, R. Hanlon, T. Seifried et al., "Microorganisms collected from the surface of freshwater lakes using a drone water sampling system (DOWSE)," *Water*, vol. 11, no. 1, p. 157, 2019.
- [207] A. Demario, P. Lopez, E. Plewka et al., "Water plume temperature measurements by an unmanned aerial system (UAS)," Sensors, vol. 17, no. 2, p. 306, 2017.
- [208] M. Messinger and M. Silman, "Unmanned aerial vehicles for the assessment and monitoring of environmental

contamination: an example from coal ash spills," *Environmental Pollution*, vol. 218, pp. 889–894, 2016.

- [209] E. Donnay, "Use of unmanned aerial vehicle (UAV) for the detection and surveillance of marine oil spills in the Belgian part of the North Sea," in *Proceedings of the 32. AMOP technical seminar on environmental contamination and response*, pp. 771–779, Canada, September 2009.
- [210] T. Chen and S. Lu, "Subcategory-aware feature selection and SVM optimization for automatic aerial image-based oil spill inspection," *IEEE Transactions on Geoscience and Remote Sensing*, vol. 55, no. 9, pp. 5264–5273, 2017.
- [211] P. Odonkor, Z. Ball, and S. Chowdhury, "Distributed operation of collaborating unmanned aerial vehicles for timesensitive oil spill mapping," *Swarm and Evolutionary Computation*, vol. 46, pp. 52–68, 2019.
- [212] Z. Ghorbani and A. H. Behzadan, "Monitoring offshore oil pollution using multi-class convolutional neural networks," *Environmental Pollution*, vol. 289, Article ID 117884, 2021.
- [213] S. Kaviri, A. Tahsiri, and H. D. Taghirad, "A cooperative control framework of multiple unmanned aerial vehicles for dynamic oil spill cleanup," *Journal of the Brazilian Society of Mechanical Sciences and Engineering*, vol. 43, no. 6, p. 289, 2021.
- [214] J. Sancho Martínez, Y. B. Fernández, P. Leinster, and M. R. Casado, "Combining unmanned aircraft systems and image processing for wastewater treatment plant asset inspection," *Remote Sensing*, vol. 12, no. 9, p. 1461, 2020.
- [215] L. Wong, B. S. Vien, Y. Ma et al., "Remote monitoring of floating covers using UAV photogrammetry," *Remote Sensing*, vol. 12, no. 7, p. 1118, 2020.
- [216] J. Burgués, M. D. Esclapez, S. Doñate, L. Pastor, S. Marco, and M. João Costa, "Aerial mapping of odorous gases in a wastewater treatment plant using a small drone," *Remote Sensing*, vol. 13, no. 9, p. 1757, 2021.
- [217] O. Panasiuk, A. Hedström, J. Marsalek, R. M. Ashley, and M. Viklander, "Contamination of stormwater by wastewater: a review of detection methods," *Journal of Environmental Management*, vol. 152, pp. 241–250, 2015.
- [218] M. Geraeds, T. V. Emmerik, R. D. Vries, and M. S. B. Razak, "Riverine plastic litter monitoring using unmanned aerial vehicles (UAVs)," *Remote Sensing*, vol. 11, no. 17, p. 2045, 2019.
- [219] C. Martin, S. Parkes, Q. Zhang, X. Zhang, M. F. McCabe, and C. M. Duarte, "Use of unmanned aerial vehicles for efficient beach litter monitoring," *Marine Pollution Bulletin*, vol. 131, pp. 662–673, 2018.
- [220] C. Ferrara, M. Lega, G. Fusco, P. Bishop, and T. Endreny, "Characterization of terrestrial discharges into coastal waters with thermal imagery from a hierarchical monitoring program," *Water*, vol. 9, no. 7, p. 500, 2017.
- [221] A. Deidun, A. Gauci, S. Lagorio, and F. Galgani, "Optimising beached litter monitoring protocols through aerial imagery," *Marine Pollution Bulletin*, vol. 131, pp. 212–217, 2018.
- [222] G. Gonçalves, U. Andriolo, L. Pinto, and D. Duarte, "Mapping marine litter with Unmanned Aerial Systems: a showcase comparison among manual image screening and machine learning techniques," *Marine Pollution Bulletin*, vol. 155, Article ID 111158, 2020.
- [223] L. Fallati, A. Polidori, C. Salvatore, L. Saponari, A. Savini, and P. Galli, "Anthropogenic Marine Debris assessment with Unmanned Aerial Vehicle imagery and deep learning: a case study along the beaches of the Republic of Maldives," *Science* of the Total Environment, vol. 693, Article ID 133581, 2019.

- [224] K. Kylili, I. Kyriakides, A. Artusi, and C. Hadjistassou, "Identifying floating plastic marine debris using a deep learning approach," *Environmental Science & Pollution Research*, vol. 26, no. 17, pp. 17091–17099, 2019.
- [225] K. Kylili, C. Hadjistassou, and A. Artusi, "An intelligent way for discerning plastics at the shorelines and the seas," *Environmental Science & Pollution Research*, vol. 27, no. 34, pp. 42631–42643, 2020.
- [226] O. Bukin, D. Proschenko, D. Korovetskiy, A. Chekhlenok, V. Yurchik, and I. Bukin, "Development of the artificial intelligence and optical sensing methods for oil pollution monitoring of the sea by drones," *Applied Sciences*, vol. 11, no. 8, p. 3642, 2021.
- [227] V. V. Klemas, "Coastal and environmental remote sensing from unmanned aerial vehicles: an overview," *Journal of Coastal Research*, vol. 315, no. 5, pp. 1260–1267, 2015.
- [228] N. Barkóczi, "Examples from the boundaries of geographic survey: architecture and flood modeling," *Small Fly. Drones Appl. Geogr. Obs*, Springer, Berlin, Germany, pp. 127–156, 2017.
- [229] C. G. David, N. Kohl, E. Casella, A. Rovere, P. Ballesteros, and T. Schlurmann, "Structure-from-Motion on shallow reefs and beaches: potential and limitations of consumergrade drones to reconstruct topography and bathymetry," *Coral Reefs*, vol. 40, no. 3, pp. 835–851, 2021.
- [230] S. S. Young and P. Wamburu, "Comparing drone-derived elevation data with air-borne LiDAR to analyze coastal sea level rise at the local level," *Papers in Applied Geography*, vol. 7, no. 3, pp. 331–342, 2021.
- [231] M. Jaud, C. Delacourt, N. Le Dantec et al., "Diachronic UAV photogrammetry of a sandy beach in brittany (France) for a long-term coastal observatory," *ISPRS International Journal* of Geo-Information, vol. 8, no. 6, p. 267, 2019.
- [232] A. Papakonstantinou, K. Topouzelis, and G. Pavlogeorgatos, "Coastline zones identification and 3D coastal mapping using UAV spatial data," *ISPRS International Journal of Geo-Information*, vol. 5, no. 6, p. 75, 2016.
- [233] D. Long, C. McCarthy, and T. Jensen, "Row and water front detection from UAV thermal-infrared imagery for furrow irrigation monitoring," in *Proceedings of the IEEE/ASME Int. Conf. Adv. Intell. Mechatronics, AIM*, pp. 300–305, Canada, September 2016.
- [234] C. Zeng, D. J. King, M. Richardson, and B. Shan, "Fusion of multispectral imagery and spectrometer data in UAV remote sensing," *Remote Sensing*, vol. 9, no. 7, p. 696, 2017.
- [235] F. Xu, Z. Gao, W. Shang et al., "Validation of MODIS-based monitoring for a green tide in the Yellow Sea with the aid of unmanned aerial vehicle," *Journal of Applied Remote Sensing*, vol. 11, no. 1, Article ID 012007, 2017.
- [236] Q. Feng, J. Liu, and J. Gong, "Urban flood mapping based on unmanned aerial vehicle remote sensing and random forest classifier—a case of yuyao, China," *Water*, vol. 7, no. 12, pp. 1437–1455, 2015.
- [237] M. Masný, K. Weis, M. Biskupič, J. González, L. M. González, and D. Santos, "Application of fixed-wing UAV-based photogrammetry data for snow depth mapping in alpine conditions," *Drones*, vol. 5, no. 4, p. 114, 2021.
- [238] S. K. Dewali, K. Jain, S. Dhamija, and D. K. Singh, "Mapping snow depth and spatial variability using SfM photogrammetry of UAV images over rugged mountainous regions of the western himalaya," *Geocarto International just-accepted*, pp. 1–26, 2022.
- [239] H. Riihimäki, M. Luoto, and J. Heiskanen, "Estimating fractional cover of tundra vegetation at multiple scales using

unmanned aerial systems and optical satellite data," *Remote Sensing of Environment*, vol. 224, pp. 119–132, 2019.

- [240] D. Yang, R. Meng, B. D. Morrison et al., "A multi-sensor unoccupied aerial system improves characterization of vegetation composition and canopy properties in the arctic tundra," *Remote Sensing*, vol. 12, no. 16, p. 2638, 2020.
- [241] A. Kartoziia, "Assessment of the ice wedge polygon current state by means of UAV imagery analysis (samoylov island, the lena delta)," *Remote Sensing*, vol. 11, no. 13, p. 1627, 2019.
- [242] D. R. Pool and J. H. Eychaner, "Measurements of aquiferstorage change and specific yield using gravity surveys," *Ground Water*, vol. 33, no. 3, pp. 425–432, 1995.
- [243] P. Brunner, H. J. Hendricks Franssen, L. Kgotlhang, P. Bauer-Gottwein, and W. Kinzelbach, "How can remote sensing contribute in groundwater modeling?" *Hydrogeology Journal*, vol. 15, no. 1, pp. 5–18, 2007.
- [244] Y. Vasuki, E. J. Holden, P. Kovesi, and S. Micklethwaite, "Semi-automatic mapping of geological Structures using UAV-based photogrammetric data: an image analysis approach," *Computers & Geosciences*, vol. 69, pp. 22–32, 2014.
- [245] Y. Vasuki, E. J. Holden, P. Kovesi, and S. Micklethwaite, "An interactive image segmentation method for lithological boundary detection: a rapid mapping tool for geologists," *Computers & Geosciences*, vol. 100, pp. 27–40, Mar. 2017.
- [246] A. Yamaguchi, R. Fukuchi, M. Hamahashi, and M. Shimizu, "UAV-based mesoscale lithologic distribution map of a large shear zone in Jurassic accretionary complex (Ohwaki outcrop in the Mino Belt, central Japan)," *Island Arc*, vol. 25, no. 6, pp. 436–438, 2016.
- [247] H. Chao, M. Baumann, A. Jensen et al., "Band-reconfigurable multi-UAV-based cooperative remote sensing for real-time water management and distributed irrigation control," *IFAC Proceedings Volumes*, vol. 41, no. 2, pp. 11744–11749, 2008.
- [248] A. J. Perea-Moreno, M. J. Aguilera-Ureña, J. E. Meroño-De Larriva, and F. Manzano-Agugliaro, "Assessment of the potential of UAV video image analysis for planning irrigation needs of golf courses," *Water*, vol. 8, no. 12, p. 584, 2016.
- [249] A. A. Kulkarni and R. Nagarajan, "Drone survey facilitated weeds assessment and impact on hydraulic efficiency of canals," *ISH Journal of Hydraulic Engineering*, vol. 27, no. 2, pp. 117–122, 2018.
- [250] J. Brinkhoff and J. Hornbuckle, "Assessment of aquatic weed in irrigation channels using UAV and satellite imagery," *Water*, vol. 10, 2018.
- [251] B. K. R. Kadapala, K. A. Hakeem, K. Raghavendra, S. Patel, and K. P. Kumar, "Capacity estimation of irrigation tanks through remote sensing from UAV platform," *J. Indian Soc. Remote Sens.*vol. 48, no. 10, pp. 1403–1411, 2020.
- [252] S. Thomson, "Thermal imaging: from irrigation management to finding leaks at the levee," *Resources Magazine*, vol. 22, pp. 10–13, 2015.
- [253] T. Chen, H. He, D. Li, P. An, and Z. Hui, "Damage signature generation of revetment surface along urban rivers using UAV-based mapping," *ISPRS International Journal of Geo-Information*, vol. 9, no. 4, p. 283, 2020.
- [254] E. Ridolfi, G. Buffi, S. Venturi, and P. Manciola, "Accuracy analysis of a dam model from drone surveys," *Sensors*, vol. 17, no. 8, p. 1777, 2017.
- [255] G. Buffi, P. Manciola, S. Grassi, M. Barberini, and A. Gambi, "Survey of the Ridracoli Dam: UAV-based photogrammetry and traditional topographic techniques in the inspection of vertical structures," *Geomatics, Natural Hazards and Risk*, vol. 8, no. 2, pp. 1562–1579, 2017.

- [256] A. Khaloo, D. Lattanzi, A. Jachimowicz, and C. Devaney, "Utilizing UAV and 3D computer vision for visual inspection of a large gravity dam," *Front. Built Environ*, vol. 4, p. 31, 2018.
- [257] A. Pijl, M. Tosoni, G. Roder, G. Sofia, and P. Tarolli, "Design of terrace drainage networks using UAV-based high-resolution topographic data," *Water*, vol. 11, no. 4, p. 814, 2019.
- [258] O. Tziavou, S. Pytharouli, and J. Souter, "Unmanned Aerial Vehicle (UAV) based mapping in engineering geological surveys: considerations for optimum results," *Engineering Geology*, vol. 232, pp. 12–21, 2018.
- [259] J. P. Duffy, A. M. Cunliffe, L. DeBell et al., "Location, location, location: considerations when using lightweight drones in challenging environments," *Remote Sens. Ecol. Conserv*, vol. 4, no. 1, pp. 7–19, 2018.
- [260] K. K. Singh and A. E. Frazier, "A meta-analysis and review of unmanned aircraft system (UAS) imagery for terrestrial applications," vol. 39, no. 15–16, p. 5078, 2018.
- [261] K. E. Joyce, S. Duce, S. M. Leahy, J. Leon, and S. W. Maier, "Principles and practice of acquiring drone-based image data in marine environments," *Marine and Freshwater Research*, vol. 70, no. 7, pp. 952–963, 2019.
- [262] M. Doukari, M. Batsaris, A. Papakonstantinou, and K. Topouzelis, "A protocol for aerial survey in coastal areas using UAS," *Remote Sensing*, vol. 11, no. 16, p. 1913, 2019.
- [263] N. Njue, J. Stenfert Kroese, J. Graf et al., "Citizen science in hydrological monitoring and ecosystem services management: state of the art and future prospects," *Science of the Total Environment*, vol. 693, Article ID 133531, 2019.
- [264] B. Weeser, J. Stenfert Kroese, S. Jacobs et al., "Citizen science pioneers in Kenya - a crowdsourced approach for hydrological monitoring," *Science of the Total Environment*, vol. 631, pp. 1590–1599, 2018.
- [265] Z. Su, Y. Zeng, N. Romano et al., "An integrative information aqueduct to close the gaps between satellite observation of water cycle and local sustainable management of water resources," *Water*, vol. 12, no. 5, p. 1495, 2020.
- [266] D. Fawcett, C. Panigada, G. Tagliabue et al., "Multi-scale evaluation of drone-based multispectral surface reflectance and vegetation indices in operational conditions," *Remote Sensing*, vol. 12, no. 3, p. 514, 2020.
- [267] C. Tomsett and J. Leyland, "Remote sensing of river corridors: a review of current trends and future directions," *River Research and Applications*, vol. 35, no. 7, pp. 779–803, 2019.
- [268] R. Kuntz Rangel, J. L. Freitas, and V. Antonio Rodrigues, "Development of a multipurpose hydro environmental tool using swarms, UAV and USV," 2019 IEEE Aerospace Conference, 2019.
- [269] C. Singh, V. Mishra, H. Harshit, K. Jain, and M. Mokros, "Application of UAV swarm semi-autonomous system for the linear photogrammetric survey," *Remote Sensing and Spatial Information Sciences*, vol. 43, pp. 407–413, 2022.
- [270] D. J. Hill and M. Babbar-Sebens, "Promise of UAV-assisted adaptive management of water resources systems," *Journal of Water Resources Planning and Management*, vol. 145, no. 7, Article ID 02519001, 2019.



Research Article

Rainfall & Seismological Dump Slope Stability Analysis on Active Mine Waste Dump Slope with UAV

Shirshendu Layek, Vasanta Govind Kumar Villuri, Radhakanta Koner, and Kapoor Chand

Department of Mining Engineering, Indian Institute of Technology (ISM), Dhanbad 826004, India

Correspondence should be addressed to Shirshendu Layek; shirshendu@me.ism.ac.in and Vasanta Govind Kumar Villuri; vgkvilluri@iitism.ac.in

Received 25 February 2022; Revised 26 May 2022; Accepted 17 June 2022; Published 12 August 2022

Academic Editor: Kamal Jain

Copyright © 2022 Shirshendu Layek et al. This is an open access article distributed under the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Mine waste dump material has no economic value to the industry. Therefore, the mine waste is dumped, forming slopes. Mine waste dump slopes obtain 30% to 50% of the mining area. To reduce the land occupancy of these slopes, they are created with high altitudes. Hence, they are susceptible to failure. Slope stability analysis is a major aspect of geotechnical engineering. Slope stability analyses are mostly done with assumptions on the geometry. This is avoided in this paper with the usage of UAVs. The 3D model is created from UAV imagery of a coal mine in Raniganj coalfield, India. The model is fine-tuned with the DGPS survey. Geotechnical data were collected and tested in the laboratory for various numerical analyses. An active mine waste dump slope is analyzed for slope failure. Earthquakes and rainfall cannot be controlled, and their effects on the stability of the mine waste dump slope were examined. The study extracted various factor of safety (FOS) analyses on static, seismic, and rainfall conditions. The seismic condition simulates a condition of the slope to be failed with low (0.948) FOS. However, rainfall condition predicts the slope to be more stable. The deformation pattern and magnitude of the slope failure are also discussed.

1. Introduction

Open-pit mining operation excavates the top portion of the soil to extract minerals or coals. During this process, the excavated soil had no economic value to the mining industry. These materials are known as waste [1, 2]. The limitation of the mining area makes it tedious to manage this waste. Therefore, the mine waste is dumped, forming a slope [3, 4]. These dump slopes are enormous. Mine waste dump is made very rapidly, therefore less time for consolidation of waste materials. This hasty approach is making the slopes susceptible to form disasters. From 1921, there were 23 accidents due to mine overburden slope failures only in India. These accidents cost 143 causalities [5, 6]. In 2016, a catastrophic slope failure in Rajmahal opencast mine caused the death of 23 people. Talcher, Bokaro, and Bharatpur recently faced similar accidents [7-9]. Mine dump slope stability analysis is a

major subject in geotechnical engineering. The safety study is based on these slopes with SSR (Shear Strength Reduction). In this technique, soil material's strength characteristics are reduced till the failure is obtained in the geometry [10]. This method provides a numeric value that represents the factor of safety [11]. If the value is greater than one, then the stability of the slope is high and viceversa [12-15]. For traditional numerical simulation, an assumption on geometry is considered [16]. With the usage of UAV, two significant problems in numerical simulations can be solved, the first reduction in geometry assumption, and the second, the GIS tools can be used for mapping, monitoring, and targeting the risk-prone areas for numerical simulation resulting in accurate analysis and less computational time. This approach is possible with UAV's RGB (red, green, and blue) imagery. Overlapped images of UAVs can create a 3D point cloud of an area [17-19]. Furthermore, DTM for slope targeting for



FIGURE 1: Map showing the study area, located in India's eastern part.

numerical simulation and geometry for numerical simulation can be developed. For testing this approach, Raniganj coalfield, India, has been selected (Figure 1 represents the study area). Raniganj is situated in the eastern part of India. This coalfield region is about 1130 km^2 from $23^\circ 25' \text{N}$ to $23^\circ 50' \text{N}$ latitude and $86^\circ 38' \text{E}$ to $87^\circ 20' \text{E}$ longitude [20].

Raniganj coal mines have most of the open cast projects of Eastern Coalfield Ltd., India. In Raniganj coalfield, the biggest open cast mine is situated in Sonepure Bazari [2]. The production of the coalfield is 33.90 million tonnes; thus, high mining activity occurs consistently. The mine we selected has an in-pit waste dumping system, i.e., waste is dumped within the mining area due to the limited range of available land. In-pit waste dump of the study area is shown in Figure 2.

An active mine dumping area was taken for testing for stability analysis, as the chances of failure will be high due to less time for waste to consolidate [12]. Hence, traditional



FIGURE 2: In-pit mining waste dump at the study area.

numerical simulation is insufficient in this scenario as rainfall plays a vital role in slope failures and earthquakes [21]. As Raniganj is a subtropical place, heavy rain occurs, and precipitation of 200 to 400 mm every monsoon has been reported yearly. Also, regular cyclones have been impeding [22]. Therefore, an increase in rain directly affects the safety factor of the overburden slope [23].

2. Materials and Methods

This study focuses on the stability analysis of the active in-pit mine waste dump. Therefore, in data acquisition, UAV imagery, DGPS survey, and geotechnical samples were collected from the study area. With the help of UAV imagery,

3D modeling and proper geometry are extracted from the active waste dump slope. The area was cross-checked with GIS to ensure the proneness to failure. Finally, three geometry of the section was extracted, and numerical simulations were done with the help of geotechnical data. In Figure 3, the steps can be followed. There are four steps to achieving the desired goal; data collection, geotechnical analysis, image processing, and numerical simulation. The numerical simulation focuses on rainfall effects on the stability of the slope and the simulation of seismological impact on the slope. These parts are discussed in this section.

2.1. UAV Data Acquisition. In the field, DJI Phantom 4 Pro UAV has been used (Figure 4 shows the UAV). The UAV can fly for up to 19 minutes on a single charge and capture 20-megapixel resolution worth of imagery. Acquisition of images is made with overlapping of 75% in horizontal and 65% in vertical. DGPS (differential global positioning system) was also used for the accuracy of the generated model [24–26].

The average GSD of the constructed orthoimagery was 5.022 cm per pixel. A total of 561 images were captured with two flights. And the flight plan was grid in nature. The area captured by the UAV is 4716900 m². DGPS can extract coordinates of an unknown area without reference. Prominent "+" symbols were drawn in the field with whitewash for this study, as shown in Figure 5. Seven GCPs were used to process the images in Pix4DMapper. As per the software Pix4DMapper with seven GCPs, the RMS should be less than 0.10, making the model highly accurate.

The overlapping helps in recovering lost 3D information of the camera with the same feature extraction in multiple images, and triangulation is done to retrieve the position of the camera in x, y, and z coordinates by reversing the approach after extracting camera location, x, y, and z coordinates of features are extracted. This approach is made by structure from motion pipeline for generating 3D point cloud and other forms of data [27, 28].

2.2. Geotechnical Analysis. Accurate shear strength parameters, i.e., cohesion and internal frictional angle of the waste materials, are necessary for numerical simulation accuracy [29, 30], along with those compaction characteristics that are also needed. For shear strength parameters, triaxial tests were performed [31], and compaction tests were carried out according to Indian Standard (IS:2720) part-6 (1965). Hence, a geotechnical investigation was necessary. The materials were initially tested with grain size distribution. According to Indian Standard Soil Classification

System (ISSCS) with various tests, the material showed coarse materials between 80% and 98%, and sand between 2 mm and 0.425 mm was ranging from 22% to 92% and, sand and clay ranging from 3.8% to 46% (particles are less than 0.75 mm) thus making the material poorly graded sand or SP. The samples were collected during UAV data acquisition and have been processed in the institute's laboratory facilities, and the extracted geotechnical parameters can be seen in Table 1.

2.3. 3D Model of UAV Imagery. 3D modeling from UAV imagery is based on epipolar geometry of photogrammetry [28, 32]. Therefore, feature extraction and matching of multiple images are done. Matched features of image pairs are used to create pose estimation (3D location of the camera of the same feature found in various images) with triangulation hence, spatial 3D point cloud data [33, 34]. To minimize errors, bundle adjustment is applied. Bundle adjustment is used on the accumulated 3D model to refine and prevent noise in triangulation [35]. This model is created from a UAV. This 3D model, along with an orthomosaicked image, was used to create a digital elevation model (DEM) for inspection of AOI.

Furthermore, to be accurate, the geometry DGPS surveyed data were integrated before processing the 3D point cloud as ground control points in Pix4DMapper. Figure 6 shows the 3D model of the study area, which is generated from UAV imagery.

2.4. Geometry Extraction for Numerical Analysis. The surveyed area from the UAV mainly covered the mining dump, but most of the area was not visible from the ground survey or observation. So, removing fewer risk-prone zones from the surveyed area helps reduce computational time. The authors targeted an active waste dump slope to be examined. A newly made dump has less time to settle down, and having loose materials causes low shear strength. Therefore, failure can occur.

In Figure 7, the active dumping zone is shown. This area initially needs to be examined with GIS. The GIS technology can cross-check whether the active dumping slope is prone to failure. This is done by developing a slope map of the study area from UAV data. Slope angles represent the relative angle displacements. In Figure 8, the slope map is shown.

From Figure 8, it can be seen that the angular value of the targeted zone exceeds 36 degrees. Thus, this site makes it ideal for examining numerical simulation for slope stability.

Three sections from the active dumping zone have been extracted. These geometries can be seen in Figures 9(b)-9(d). The geometries from the same dumping slope have different angles and heights. Examining only one will derive less in-depth analysis. These sections have a length of 207, 210, and 198 meters horizontally and 66, 83, and 101 meters in altitude displacement (the projection in the figure is in 3D; hence, the perspective of displacement on different axis differs).



FIGURE 3: Slope stability analysis method in static, seismic, and rainfall conditions with the UAV.



FIGURE 4: UAV data acquisition in the study area.

3. Results and Discussion

Flac comes with computing packages for numerical simulation applications. This is widely accepted and recognized. In Flac3D, numerical analysis is done with the shear strength reduction technique or SSR. Constraint boundaries restrict developed geometries. This approach improves accuracy, efficiency, and modeling time, ensuring accuracy and legitimacy [36, 37]. Three geometries were extracted from the targeted waste dump slope. These geometries range between 66 meters and 101 meters in height and 197 meters to 210 meters horizontally. Mohr-Coulomb's failure criteria were adopted in the first iteration of stability analysis. The failure occurs with the critical combination of the shear strength of loose mine waste dump materials with normal stress. The loose, broken waste dump materials contain low shear strength [38, 39]. The dump slope was numerically analyzed with finite element method codes, and the FOS was calculated with displacement values, as shown in Table 2.

The slope seems to be stable with low displacement values in all the scenarios [38]. Furthermore, the sliding portion of the geometries can be seen in Figure 10. All the failures are circular. The FOS of the sections ranged from 1.26 to 1.53 of the slope under static conditions.

As discussed earlier, static slope stability analysis is insufficient as two significant reasons slope failures occur due to heavy rainfall and seismological effects. These two factors cannot be dealt with human intervention. Therefore, examining these two factors will ensure a more nuanced analysis on analysis.

3.1. Pseudostatic Condition for Seismic Effect. India is divided into four zones for an earthquake. The study area of this paper has zone III in terms of earthquake hazards. According to Figure 11, the study area lies in the moderate damage risk zone, and the factor is 0.16.

A seismic load is given in the numerical simulation of the seismological effect, i.e., 0.16. Therefore, the integrity of the analysis of the limit equilibrium method [41] with seismic conditions on the geometries is ensured. The extracted FOS from the simulation validated a lower stability factor than the static condition. In Table 3 no. 3 FOS of three sections can be seen.

The seismological effect also extracted similar failure patterns as a static condition. The LEM method provides only a stability factor. Therefore, stress and strain information is not achievable, resulting in nondeformation analysis in pseudostatic conditions. The sections after numerical simulation can be seen in Figure 12.





(c)

FIGURE 5: (a, b) represent temporary ground control point and (c) DGPS data acquisition for improving the accuracy of the UAV dataset.

TABLE 1: Waste dump material properties extracted from laboratory tests.

Parameters	Tested result
Soil unit weight (kN/m ³)	17.59914
Friction angle (degree)	25.556
Cohesion (Pa)	50000
Bulk modulus (GPa)	0.32
Shear (GPa)	3.1



FIGURE 6: (a, b) 3D models of the study area created from UAV imagery from different views.



FIGURE 7: Active waste dumping zone in the study area.



FIGURE 8: Slope map of the area and (b) cropped area represents the active waste dump slope.

This section validates the analysis methodology as to its capability to calculate the slope failure surface with seismic acceleration. The simulated result is documented with zone III seismic acceleration (Figure 11), which has been analyzed through widely recognized conventional Bishop's limit equilibrium methods [42]. The sections were further analyzed with a 2, 4, 6, and 8 seconds dynamic seismic load.

The el Centro dataset is adopted for input load. The given input data can be seen in Figure 13. This data propagates in

the N-S direction. The pick of the acceleration is 0.3 g recorded around 2 seconds, and after 5 seconds, the acceleration gradually decreases with a low spike at 8 seconds and in around 25 seconds with 0.138 g. The displacement of the sections with dynamic load can be seen in Figure 14. In section 3 (Figure 13(e)), the displacement initially started with negative and then moved towards positive, which strongly proposes crack generation in the upper part of section 3 [43].

Advances in Civil Engineering



FIGURE 9: (a) Three sections marked with dotted lines on the active dumping slope. (b-d) Geometry of section 1, section 2, and section 3, respectively.

TABLE 2: FOS of 3 sections in static numerical simulation condition with deformation.

	FOS	Deformation (meters)
Section 1	1.53	1.059E - 01
Section 2	1.38	8.084E - 02
Section 3	1.26	2.226 <i>E</i> – 01







(c)

FIGURE 10: (a-c) circular failure profiles of sections 1, 2, and 3.



FIGURE 11: Seismological map of India [40].

	FOS
Section 1	1.219
Section 2	1.048
Section 3	0.953





FIGURE 12: Pseudostatic analysis of three sections.











FIGURE 14: Dynamic load of seismic data with 2, 4, 6, and 8 seconds of sections 1, 2, and 3.

The simulation shows that the FOS of section 3 has a value of 0.953. This makes the section prone to failure. As for sections 1 and 2, they are shown a dip in FOS compared to numerical simulation in static conditions with 1.219 and 1.048.

3.2. Dump Slope Stability Analysis with Average Rainfall. The study area is subtropical in climatic nature. Hence, the rainfall occurs heavily after summer, i.e., from June to September [44–46]. Mine waste dump failures occur primarily in this period due to heavy precipitation. The precipitation in the study area was highest in August. In August, the minimum rainfall was 207.3 millimeters recorded in 2017, and the highest was in 2016 with 601.3 millimeters. In other monsoon months, precipitation varies from 28 millimeters to 515 millimeters. The average rainfall in the monsoon of the study area has been recorded as 264.266 mm in the last ten years. The numerical analysis of the study area with precipitation data was simulated for seven months. Extracted FOS can be seen in Table 4. The FOS in this scenario improved [47].

The improvement in FOS is explainable by two factors. First, seven months for settlement of the slope which is after

TABLE 4: FOS with average rainfall during monsoon.

	FOS	Deformation (centimeters)
Section 1	1.60	7.474
Section 2	1.40	14.4
Section 3	1.25	109.8

the monsoon. The second cause is related to the waste dump material characteristic. The material contains clay that has high adhesiveness when mixed with water, which vastly improves the cohesion that benefits the bonding and stability [48]. The saturation profile of the simulation can be seen in Figure 15. This phenomenon was derived from the theory of consolidation, in which precipitation decreases the pour spaces of sand and clay-type materials and improves the consolidation with time.

Deformation of the sections stayed almost the same with low movements of the slope materials, which maxed at 1.098 meters at section 3. The displacement profile of the sections can be seen in Figure 16. The displacement on the slopes with rainfall causes less than 1 meter of negative deformation. In sections 1 and 2, the displacement goes up to 0.05 meters and



FIGURE 15: (a-c) represent the saturation profiles of section 1, section 2, and section 3, respectively, with a flow time of 7 months.

0.16 meters, although insection 3 geometry, the deformation goes up to 1 meter.

As a future study, rigorous analyses can be done with various FORM algorithms, and these algorithms were

derived from HLRF algorithms. HLRF and FORM algorithms are known for their reliability [49, 50]. Therefore, a basic comparison of the stability of waste dump slopes can be compared with various material properties [51].



FIGURE 16: (a-c) represent the displacement profile in meters. The plotted displacement of the X and Y axes along the surface of the geometries of sections 1, 2, and 3 from toe to the pick of the sections.

4. Conclusions

The availability of the UAV and improvement on the image sensor is useful for creating 3D maps and models of the mining area. Proper scale and dimensions are extracted with the help of UAV imagery, DGPS, and photogrammetry. The geometries extracted from the UAV's 3D model represent an active in-pit mine dump slope, which is examined based on less time for the slope to settle waste materials. A geotechnical investigation was carried out on the dump waste materials. The geotechnical properties are instrumental in numerical analysis reliability. Numerical simulations on these geometries are carried out in three different approaches, making the study cohesive. The environments for the numerical simulation consisted of static, seismic, and rainfall conditions. These simulations show the FOS value initially at a stable state, and rainfall with seven months of simulation time increased the stability. However, the seismic condition in Bishop's limit equilibrium method shows significantly lower FOS of the sections. The least FOS was 0.948 achieved in section 3 geometry after seismic load. This outcome illustrates the failure in section 3.

Nevertheless, regular precipitation and time (or settlement of the loose materials) suggest a higher FOS, so the waste dump slope will be stable if no significant seismic acceleration occurs. The displacement analysis also indicates a maximum of 1-meter deformation. Although with seismic activity, cracks will be formed in section 3.

Data Availability

The data that support the findings of this study are available on request from the corresponding author. The data are not publicly available due to privacy or ethical restrictions.

Conflicts of Interest

The authors declare no conflicts of interest.

Acknowledgments

The research is funded by the Science and Engineering Research Board (SERB), New Delhi (Grant No - CRG/2020/ 005919).

References

- H. B. Sahu and S. Dash, "Land Degradation Due to Mining in India and its Mitigation Measures," in *Proceedings of the 2011* 2nd International Conference on Environmental Science and Technology IPCBEE, Singapore, 2011.
- [2] I. Roy: Slope Stability Study of External Dump of Sonepur-Bazari Slope Stability Study of External Dump of Sonepur-Bazari Opencast Coal Mine, India-A Case Study Opencast Coal Mine, India-A Case Study SLOPE STABILITY STUDY of EXTERNAL DUMP of SONEPUR-BAZARI OPENCAST COAL MINE, INDIA-A CASE STUDY. Delhi, India, 2008.
- [3] T. Gvozdkova, P. Koščová, and M. Koščová, "On the measures to solve environmental problems of the territory with

intensive open pit coal mining," E3S Web of Conferences, vol. 105, Article ID 02006, 2019.

- [4] K. Pactwa, J. Dudek, and M. Dudek, "Coal mining waste in Poland in reference to circular economy principles," *Fuel*, vol. 270, Article ID 117493, 2020.
- [5] A. K. Dash, "EBSCOhost |138948870| Analysis of accidents due to slope failure in Indian opencast coal mines," 2022, https://web.p.ebscohost.com/abstract?direct=true&profile= ehost&scope=site&authtype=crawler&jrnl=00113891&AN= 138948870&h=uO0Bs3%2FvzVASnZvnzdz2PLiJf5xD7QX9 SjsZvQOBL%2BEchP3m6CImzVtu57%2Bppidd1AjIVmu0V1 RCUbuhIP3m4Q%3D%3D&crl=c&resultNs=AdminWeb Auth&resultLocal=ErrCrlNotAuth&crlhashurl=login.aspx% 3Fdirect%3Dtrue%26profile%3Dehost%26scope%3Dsite% 26authtype%3Dcrawler%26jrnl%3D00113891%26AN%3D138 948870.
- [6] M. F. Labib, A. S. Rifat, M. M. Hossain, A. K. Das, and F. Nawrine, "Road accident Analysis and Prediction of accident Severity by using machine Learning in Bangladesh," in *Proceedings of the 2019 7th Int. Conf. Smart Comput. Commun. ICSCC 2019*, Sarawak, Malaysia, June 2019.
- [7] Down to Earth, "Talcher coal mine accident: 1 dead, 4 trapped after landslide," 2019, https://www.downtoearth.org.in/news/ mining/talcher-coal-mine-accident-1-dead-4-trapped-afterlandslide-65811.
- [8] A. Abhishek, "Four trapped in Bokaro mine for over 30 hours come out safely on their own | India News, The Indian Express," 2021, https://indianexpress.com/article/india/fourtrapped-in-bokaro-mine-for-over-30-hours-come-out-safelyon-their-own-7648261/.
- [9] "4th body found from odisha coal mine accident site, manager suspended for negligence," 2019, https://www.news18.com/ news/india/4th-body-found-from-odisha-coal-mine-accidentsite-manager-suspended-for-negligence-2248703.html.
- [10] H. Lu, L. Xu, and D. Fredlund, "Comparison of 3D Finite Element Slope Stability with 3D Limit Equilibrium Analysis," in Proceedings of the 18th International Conference on Soil Mechanics and Geotechnical Engineering, Paris, France, September 2013.
- [11] X. Fu, Q. Sheng, Y. Zhang, J. Chen, S. Zhang, and Z. Zhang, "Computation of the safety factor for slope stability using discontinuous deformation analysis and the vector sum method," *Computers and Geotechnics*, vol. 92, pp. 68–76, 2017.
- [12] E. M. Dawson, W. H. Roth, and A. Drescher, "Slope Stability Analysis by Strength Reduction," *Géotechnique*, vol. 49, pp. 835–840, 2015.
- [13] Y. M. Cheng, T. Wei, and W. B. Wei, "Two-dimensional slope stability analysis by limit equilibrium and strength reduction methods," *Computers and Geotechnics*, vol. 34, no. 3, pp. 137–150, 2007.
- [14] J. M. Duncan, "State of the art: limit equilibrium and finiteelement analysis of slopes," *Journal of Geotechnical Engineering*, vol. 122, no. 7, pp. 577–596, 1996.
- [15] N. Bar, M. Kostadinovski, M. Tucker et al., "Rapid and robust slope failure appraisal using aerial photogrammetry and 3D slope stability models," *International Journal of Mining Science and Technology*, vol. 30, no. 5, pp. 651–658, 2020.
- [16] A. J. Li, R. S. Lyamin, and A. V. Lyamin, "Limit analysis solutions for three dimensional undrained slopes," *Computers* and Geotechnics, vol. 36, no. 8, pp. 1330–1351, 2009.
- [17] S. Ullman, "The interpretation of structure from motion," Proceedings of the Royal Society of London, Series B: Biological Sciences, vol. 203, pp. 405–426, 1979.

- [18] J. Jensen, A. Mathews, J. L. R. Jensen, and A. J. Mathews, "Assessment of image-based point cloud products to generate a bare earth surface and estimate canopy heights in a woodland ecosystem," *Remote Sensing*, vol. 8, no. 1, p. 50, 2016.
- [19] S. Layek, R. K. Singh, V. G. K. Villuri, R. Koner, A. Khare, and R. Khare, "3D reconstruction: an emerging prospect for surveying," *Applications of Geomatics in Civil Engineering*, , pp. 71 • 81, Springer, Singapore, , 2020.
- [20] S. Murthy, B. Chakraborti, and M. D. Roy, "Palynodating of Subsurface Sediments, Raniganj Coalfield, Damodar Basin, West Bengal," *Journal of Earth System Science*, vol. 119, 2010.
- [21] W. Du, Q. Sheng, X. Fu, J. Zhou, and Y. Zhou, "Extensions of the two-phase double-point material point method to simulate the landslide-induced surge process," *Engineering Analysis with Boundary Elements*, vol. 133, pp. 362–375, 2021.
- [22] R. S. Sannasiraj and S. A. Sannasiraj, "Climate change projection of wave climate due to Vardah cyclone in the Bay of Bengal," *Dynamics of Atmospheres and Oceans*, vol. 97, Article ID 101279, 2022.
- [23] R. Chakravarty and D. Chakravarty, "Numerical analysis of rainfall effects in external overburden dump," *International Journal of Mining Science and Technology*, vol. 26, no. 5, pp. 825–831, 2016.
- [24] C. Hoppe, M. Klopschitz, M. Donoser, and H. Bischof, "Incremental Surface Extraction from Sparse Structure-From-Motion Point Clouds," in *Proceedings of the British Machine Vision Conference (BMVC)*, Bristol, UK, September 2013.
- [25] K. Douterloigne, S. Gautama, and W. Philips, "On the accuracy of 3D landscapes from UAV image data," in *Proceedings of the 2010 IEEE International Geoscience and Remote Sensing Symposium*, pp. 589–592, Honolulu, HI, USA, July 2010.
- [26] H. Chikatsu and H. Chikatsu, "Camera calibration in 3D modelling for UAV application," *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. XL-4/W5, pp. 223–226, 2015.
- [27] M. J. Westoby, J. Brasington, N. F. Glasser, M. J. Reynolds, and J. M. Reynolds, "Structure-from-Motion' photogrammetry: a low-cost, effective tool for geoscience applications," *Geomorphology*, vol. 179, pp. 300–314, 2012.
- [28] J. L. Schönberger and J.-M. Frahm, "Structure-from-Motion Revisited," in *Proceedings of the 2016 IEEE Conference on Computer Vision and Pattern Recognition (CVPR)*, Las Vegas, NV, USA, June 2016.
- [29] A. McQuillan, I. Oh, and J. Oh, "Methods applied in Australian industry to evaluate coal mine slope stability," *International Journal of Mining Science and Technology*, vol. 30, no. 2, pp. 151–155, 2020.
- [30] L. Zdravković and D. M. Potts, "Application of numerical analysis in geotechnical engineering practice," *GeoFlorida*, pp. 69–88, 2010.
- [31] Y. Zhou, Q. Sheng, N. Li, and X. Fu, "The dynamic mechanical properties of a hard rock under true triaxial damage-controlled dynamic cyclic loading with different loading rates: a case study," *Rock Mechanics and Rock Engineering*, vol. 1, pp. 1–22, 2022.
- [32] I. Jazayeri, C. Fraser, S. Cronk, I. Jazayeri, C. S. Fraser, and S. Cronk, *International Archives Of Photogrammetry, Remote Sensing and Spatial Information Sciences*, International Society for Photogrammetry and Remote Sensing, French, German, 2010.

- [33] N. Snavely, S. M. Szeliski, and R. Szeliski, "Modeling the world from Internet photo collections," *International Journal of Computer Vision*, vol. 80, no. 2, pp. 189–210, 2008.
- [34] M. Favalli, A. Fornaciai, I. Isola, S. Nannipieri, and L. Nannipieri, "Multiview 3D reconstruction in geosciences," *Computers & Geosciences*, vol. 44, pp. 168–176, 2012.
- [35] M. R. Robson and S. Robson, "Straightforward reconstruction of 3D surfaces and topography with a camera: accuracy and geoscience application," *Journal of Geophysical Research: Earth Surface*, vol. 117, no. F3, pp. a–n, 2012.
- [36] P. Zou, X. Zhao, Z. Meng, A. Li, Z. Hu, and W. Hu, "Sample rocks tests and slope stability analysis of a mine waste dump," *Advances in Civil Engineering*, vol. 2018, pp. 1–17, 2018.
- [37] H. Lin, P. Cao, J. T. Li, X. L. Jiang, and Z. M. He, "Automatic generation of FLAC3D model based on SURPAC," *Journal of China University of Mining and Technology*, vol. 37, no. 3, pp. 339–342, 2008.
- [38] A. Kainthola, D. Verma, S. S. Singh, and T. N. Singh, "A coal mine dump stability analysis-A case study," *Geomaterials*, vol. 1, no. 1, pp. 1–13, 2011.
- [39] R. Chakravarty and D. Chakravarty, "Characterisation of overburden dump materials: a case study from the Wardha valley coal field," *Bulletin of Engineering Geology and the Environment*, vol. 75, no. 3, pp. 1311–1323, 2016.
- [40] W. K. Verma and A. K. Verma, "Probabilistic seismic hazard analysis for Kakrapar atomic power station, Gujarat, India," *Natural Hazards*, vol. 69, no. 1, pp. 919–952, 2013.
- [41] R. Chakravarty and D. Chakravarty, "Earthquake response of external mine overburden dumps: a micromechanical approach," *Natural Hazards*, vol. 56, no. 3, pp. 941–959, 2011.
- [42] J. Ji, W. Zhang, F. Zhang, Y. Lü, and Q. Lü, "Reliability analysis on permanent displacement of earth slopes using the simplified bishop method," *Computers and Geotechnics*, vol. 117, Article ID 103286, 2020.
- [43] R. Koner, "D.C.-A. in G.E. and undefined," Evaluation of Seismic Response of External Mine Overburden Dumps, researchgate.net, 2010.
- [44] R. Chakrabortty, S. C. Pal, S. Das, and B. Das, "Modeling and mapping of groundwater potentiality zones using AHP and GIS technique: a case study of Raniganj Block, Paschim Bardhaman, West Bengal," *Modeling Earth Systems and Environment*, vol. 4, no. 3, pp. 1085–1110, 2018.
- [45] A. K. Singh, M. K. Mahato, B. Singh, and K. K. Singh, "Quality assessment of mine water in the Raniganj coalfield area, India," *Mine Water and the Environment*, vol. 29, no. 4, pp. 248–262, 2010.
- [46] X. Fu, Q. Sheng, G. Li, Z. Zhang, Y. Du, and Y. Du, "Analysis of landslide stability under seismic action and subsequent rainfall: a case study on the Ganjiazhai giant landslide along the Zhaotong-Qiaojia road during the 2014 Ludian earthquake, Yunnan, China," *Bulletin of Engineering Geology and the Environment*, vol. 79, no. 10, pp. 5229–5248, 2020.
- [47] F. q. Tang, "Mining subsidence monitoring using the method of combining InSAR and GPS technology," *Journal of Coal Science and Engineering*, vol. 17, no. 2, pp. 133–136, 2011.
- [48] G. Erbach and D. C. Erbach, "Effect of wetting and drying on soil physical properties," *Journal of Terramechanics*, vol. 36, no. 1, pp. 39–49, 1999.
- [49] J. Ji, C. Zhang, Y. Kodikara, and J. Kodikara, "Reliability-based design for geotechnical engineering: an inverse FORM

approach for practice," Computers and Geotechnics, vol. 111,

- pp. 22–29, 2019. [50] J. Ji, C. Zhang, Y. Kodikara, and J. Kodikara, "Effect of 2D spatial variability on slope reliability: a simplified FORM analysis," Geoscience Frontiers, vol. 9, no. 6, pp. 1631-1638, 2018.
- [51] J. Ji, C. W. Wang, Y. Gao, and L. Zhang, "Probabilistic investigation of the seismic displacement of earth slopes under stochastic ground motion: a rotational sliding block analysis," Canadian Geotechnical Journal, vol. 58, pp. 952-968, 2021.